Effects of Shirasu Natural Pozzolan and Limestone Powder on the Strength Development and Chemical Resistance of Concrete

(しらすならびに石灰石微粉末がコンクリートの圧縮強度および化学抵抗性

に及ぼす影響)

HIROSHIMA UNIVERSITY

March 2020

PHAM VAN TOAN

Effects of Shirasu Natural Pozzolan and Limestone Powder on the

Strength Development and Chemical Resistance of Concrete

(しらすならびに石灰石微粉末がコンクリートの圧縮強度および化学抵抗性

に及ぼす影響)

PHAM VAN TOAN

A THESIS SUBMITTED

FOR THE DEGREE OF DOCTOR OF ENGINEERING

DEPARTMENT OF CIVIL AND ENVIRONMENTAL ENGINEERING

HIROSHIMA UNIVERSITY

March 2020

ABSTRACT

Concrete is the most widely used construction material and concrete structures have been faced with a variety of aggressive environment. Recently, seawater in coastal acid sulfate soils contains many aggressive chemical compositions for concretes structure and affects huge areas in the world. In addition, the environment in some sewer systems can become very aggressive, especially in the case of sulfuric acid formed by the bacterial action or industrial products combined with magnesium chloride which is used as deicing salt in cold climate areas and drains into the sewer system. In these environments, H⁺, SO₄²⁻, Mg²⁺, Cl⁻ ions are considered as the most destructive ions for concrete structures. However, very limited studies have examined the detrimental effect of these ions on the durability of concrete which associated with the service life and maintenance cost of civil infrastructure. The present study aims at investigating the effect of Shirasu, a large deposit of pyroclastic flows in Kyushu island of Japan, and limestone powder as fine aggregate in concrete with respect to compressive strength improvement and the chemical resistance of concrete against aggressive environments containing H⁺, SO₄²⁻, Mg²⁺, Cl⁻ ions.

To achieve the above-mentioned purposes, this thesis is divided into of sixes chapters, and detailed organization of the thesis is described as follows:

Chapter 1 presents the background, purposes, and methodology of this study.

Chapter 2 provides a brief literature review on strength development and durability of concrete by considering the effects of limestone powder and Shirasu natural pozzolan.

Chapter 3 presents the experimental program consisting of materials, the mixing, and the casting procedure. In this study, Shirasu and limestone powder were used as a partial fine aggregate replacement. The replacement ratios of fine aggregate by limestone powder and Shirasu were 5% and 55% by volume, respectively, based on referring to some previous studies. After casting, specimens of each concrete mixture, including cylindrical ones with 100 mm

diameters and 200 mm heights and cubic ones with dimensions of $100 \times 100 \times 100$ mm, were separated into two groups: one was stored under normal curing temperature condition in which specimens were cured in a sealed condition with aluminum tape at 20°C until the designated test ages (i.e., 3, 28, and 91 days for the compressive strength test); whereas other was placed under sealed condition with aluminum tape at high curing temperature history of 60°C for 8 hours followed by cooling process in the curing chamber. For the immersion test, specimens cured until the age of 56 days were immersed in a mixed solution containing 3% H₂SO₄ and 5% MgCl₂. The effects of limestone powder and Shirasu on properties of concrete cured at different initial temperatures were evaluated via the measurement of compressive strength, Ca(OH)₂ content by using thermogravimetric analysis, X-ray diffraction analysis, erosion depth, chloride penetration depth, the mass loss and compressive strength reduction after immersion in the mixed solution.

Chapter 4 discusses the effects of limestone powder and Shirasu natural pozzolan on the strength development and chemical resistance of concretes cured in normal curing temperature with sealed aluminum adhesive tape through microstructural and chemical investigations. The results showed that the use of Shirasu natural pozzolan as a fine aggregate replacement can improve the strength development of concretes after 28 days curing. In addition, after 22 weeks of immersion in the mixed solution, the compressive strength loss, neutralization depth and chloride penetration depth of concretes containing Shirasu were lower than those of concretes without Shirasu. It indicated that the replacement of fine aggregate with Shirasu can improve not only the compressive strength but also the chemical resistance of concrete to the aggressive strength was insignificant. In addition, after immersion in the mixed solution, highest compressive strength loss, neutralization depth, erosion depth of concretes containing limestone

powder was attributed to the higher contents of portlandite and calcite which were vulnerable to the aggressive chemical reaction.

Chapter 5 discusses the effects of different initial curing temperatures on the strength development and chemical resistance of concrete to the mixed solution through microstructural and chemical investigations. The results demonstrated that initial high curing temperature increased the compressive strength of concrete after aging for 3 days but reduced it at later ages (i.e. 28 and 91 days) in comparison with that of concrete cured at normal temperature of 20°C. In addition, initial high curing temperature can accelerate the pozzolanic reactivity of Shirasu, leading to an increase in the compressive strength of concretes containing Shirasu. Furthermore, the deterioration of concretes cured at high temperature by the attack of the mixed solution was less severe than that of concretes cured at normal temperature.

Chapter 6 states the conclusions of this research. Recommendations for future work are also provided.

ACKNOWLEDGEMENT

First and foremost, I would like to express my sincere gratitude and appreciation to my supervisor, Professor Kenji Kawai, whose guidance continued support and provided the necessary enthusiasm for my doctoral course. I thank him for going along with me in every step of my research.

I would like to gratefully acknowledge Professor Kenichiro Nakarai for his useful comments and suggestions in my study. Throughout the English seminar which is held by Professor Kenichiro Nakarai and Assistant Professor Dr. Yuko Ogawa and other lecturers, the foreign students, including me, and Japanese students have an academic site for opening our knowledge and receiving more useful advice in our research progresses.

I wish to thank co-sub supervisor Professor Takashi Tsuchida for his kind assistance and supervision.

I would like to deeply thank Assistant Professor Dr. Yuko Ogawa for her kind support during my doctoral course whenever I need her support.

I would like to express my gratitude to Specially Appointed Professor Ryoichi Sato for his kind encouragement.

I wish to thank ex-Assistant Professor Dr. Bui Phuong Trinh and Assistant Professor Dr. Ho Si Lanh for their kind support and encouragement during my doctoral course.

I would like to express my gratitude to Assistant professor Dr. Riya Catherine George for her kind comment and support.

I particularly thank Mr. Meng Phalkong and all of the members in Zaiken laboratory for helping me in different ways in the course of this work. I must also thank the Vietnamese government scholarship for the funding support during my doctoral course. Besides, I would like to express my gratitude to Vietnam Maritime University (Viet Nam) for giving me the chance of this course.

Last, I would like to express my gratitude to my family, including my wife, my children, my parents, and my parents-in-law for their sacrifice, support, and endless love.

Hiroshima, December 2019

Pham Van Toan

CONTENTS

ABSTRACT		i
ACKNOWLED	GEMENT	iv
LIST OF FIGUR	RES	ix
LIST OF TABL	ES	xii
Chapter 1. INTR	ODUCTION	1
1.1. Genera	l background	1
1.2. Objecti	ves of the research	2
1.3. Method	lology of the research	3
1.4. Thesis	outline	4
Chapter 2. LITE	RATURE REVIEWS	8
2.1. Effect of	of limestone powder on the properties of concrete	8
2.1.1. Fil	ler effect	8
2.1.2. Nu	cleation effect	8
2.1.3. Di	lution effect	9
2.1.4. Ch	emical effect	10
2.1.5. Ef	fect of limestone powder on fresh concrete	11
2.1.6. Ef	fect of limestone powder on hardened concrete	13
2.1.6.1.	Effect of limestone powder on shrinkage	13
2.1.6.2.	Effect of limestone powder on microstructure	14
2.1.6.3.	Effect of limestone powder on mechanical properties	16
2.1.6.4.	Effect of limestone powder on durability	19
2.2. Effect of	of natural pozzolan on the properties of concrete	23
2.2.1. Ch	emical and physical properties	23
2.2.2. Re	action mechanism	24
2.2.3. Ef	fect of natural pozzolan on fresh paste/mortar/concrete	25
2.2.4. Ef	fect of natural pozzolan on hardened concrete	26
2.2.4.1.	Effect of natural pozzolan on the microstructure	26
2.2.4.2.	Effect of Shirasu natural pozzolan on the mechanical properties	
2.2.4.3.	Effect of Shirasu natural pozzolan on the durability	
2.2.5. Sh	irasu natural pozzolan	
2.3. Effect of	of curing temperature on properties of concrete	

2.4. Dete	rioration of concrete in aggressive environment	
2.4.1.	Deterioration of concrete caused by acid environment	
2.4.2.	Deterioration of concrete caused by magnesium chloride	37
2.4.3.	Some methods of improving the resistance of concrete to	aggressive
environm	nents	
2.4.3.1	. Type of cement	
2.4.3.2	. The incorporation of mineral admixture	
2.4.3.3	. Water–cement ratio	
2.5. Sum	mary	
Chapter 3. EX	XPERIMENTAL PROGRAM	
3.1. Intro	oduction	
3.2. Mate	erials	
3.3. Mixt	ture proportions	53
3.4. Test	procedures and measurements	55
3.4.1.	Specimen preparation	55
3.4.2.	Ca(OH) ₂ content	55
3.4.3.	Compressive strength	56
3.4.4.	Chloride diffusion coefficient of concrete by steady-state migration te	st57
3.4.5.	Immersion test	58
3.4.5.1	. Procedures	58
3.4.5.2	. X-ray diffraction (XRD) analysis	59
3.4.5.3	. Erosion depth	59
3.4.5.4	Compressive strength loss	60
3.4.5.5	. Neutralization depth	60
3.4.5.6	Chloride penetration depth measurement	61
3.4.5.7	Concentration profiles of magnesium, chloride and sulfate ions in	n concretes
	61	
Chapter 4. EF	FECTS OF LIMESTONE POWDER AND SHIRASU NATURAL PO	DZZOLAN
ON THE STR	RENGTH DEVELOPMENT AND CHEMICAL RESISTANCE OF CO	ONCRETE
		65
4.1. Intro	oduction	65
4.2. Expe	eriments	65
4.2.1.	Mixture proportion	65
4.2.2.	Specimen preparation and curing condition	66

4.2.3. Te	est procedures and measurements	66
4.3. Result	s and discussion	66
4.3.1. C	a(OH) ₂ content	66
4.3.2. C	ompressive strength	67
4.3.3. In	nmersion test	68
4.3.3.1.	XRD analysis	68
4.3.3.2.	Erosion depth	70
4.3.3.3.	Compressive strength loss	72
4.3.3.4.	Concentration profiles of magnesium, chloride and sulfate ions ir	1 concretes.
	73	
4.4. Conclu	usions	76
Chapter 5. EF	FECTS OF INITIAL CURING TEMPERATURE ON THE S	FRENGTH
DEVELOPME	NT AND CHEMICAL RESISTANCE OF CONCRETE CON	JTAINING
LIMESTONE F	POWDER AND SHIRASU NATURAL POZZOLAN	79
5.1. Introdu	uction	79
5.2. Experi	iments	79
5.2.1. M	lixture proportion	79
5.2.2. S ₁	pecimen preparation and curing condition	80
5.2.3. Te	est procedures and measurements	81
5.3. Result	s and discussion	81
5.3.1. C	a(OH) ₂ content	81
5.3.2. C	ompressive strength	
5.3.3. C	hloride diffusion coefficient	
5.3.4. In	nmersion test	
5.3.4.1.	XRD analysis	
5.3.4.2.	Erosion depth	
5.3.4.3.	Compressive strength loss	
5.3.4.4.	Neutralization depth	91
5.3.4.5.	Chloride penetration depth	92
5.4. Conclu	usions	93
Chapter 6. CON	ICLUSIONS AND RECOMMENDATIONS	97
6.1. Conclu	usions	97
6.2. Recon	nmendations for future study	98
Appendix: Publ	lications arising from this research	100

LIST OF FIGURES

Figures	Title
Chapter	2
2.1	Influence of particles size of limestone powder (LS) on the heat release rate
2.2	Heat of hydration of Portland cement-20 μ m limestone
2.3	XRD-patterns for the different tested blends at 1, 7, 28, 90 and 180 days
2.4	Setting time of concrete with 350kg/m ³ of cement with different limestone
	powder content
2.5	Total and autogenous shrinkages of concrete after 28 days
2.6	Pore size distribution of mixtures with 350kg/m ³ of cement
2.7	Porosity and average pore size of blended binder pastes after 28 days of
	hydration
2.8	Effects of limestone powder content on the compressive strength of mortars
2.9	Effects of limestone powder content on the water penetration depth of
	concrete
2.10	Effects of limestone powder content on the sorptivity of concrete
2.11	Total charge passed versus W/C ratio in rapid chloride penetration test
2.12	Effects of different factors on the chloride diffusion coefficient of concrete
2.13	Carbonation depth of concrete versus limestone powder content
2.14	BEI of OPCS-VA cured for 1 year at (A) 10°C and (B) 60°C
2.15	Variation of threshold radius with time and metakaolin content
2.16	Variation of water absorption of concrete containing metakaolin
2.17	Variation of electrical resistivity of concrete containing volcanic ash
2.18	Source of Shirasu
2.19	Microscopic of Shirasu particle

Chapter 3

- 3.1 Particle size distributions of the fine aggregate mixtures
 3.2 The apparatus used for steady-state migration experiments.
 3.3 Erosion depth (a), chloride penetration (b) and depth neutralization depth (c) measurements
 3.4 Sampling of a deteriorated concrete specimen to measure Cl⁻, SO₄²⁻ and Mg²⁺
- 3.4 Sampling of a deteriorated concrete specimen to measure $C\Gamma$, $SO_4^{2^2}$ and Mg^{2^+} ion concentrations

Chapter 4

- 4.1 Ca(OH)₂ content in concrete samples at different ages
- 4.2 Compressive strengths of concrete specimens after aging for 3, 28 and 91 days
- 4.3 XRD patterns of concretes immersed in the mixed solution or in deionized water for 22 weeks
- 4.4 Erosion depths of concrete specimens during immersion in the mixed solution
- 4.5 Compressive strength losses of concrete specimens after 22 weeks of immersion
- 4.6 The Cl- ion penetration profiles in various concretes after 22 weeks of immersion
- 4.7 The SO_4^{2-} ion penetration profiles in various concretes after 22 weeks of immersion
- 4.8 The Mg²⁺ ion penetration profiles in various concretes after 22 weeks of immersion The Mg²⁺ ion penetration profiles in various concretes after 22 weeks of immersion

Chapter 5

- 5.1 Schematic representation of the heat treatment procedure
- 5.2 Ca(OH)₂ content in cement pastes of concretes cured at different curing temperatures

- 5.3 Compressive strengths of concrete specimens after aging for 3, 28 and 91 days
- 5.4 Cumulative increase of chloride concentration in anode sections
- 5.5 Chloride diffusion coefficients of concretes
- 5.6 XRD patterns of concretes after 22 weeks of immersion in water and mixed solution
- 5.7 Erosion depth of concrete after 22 weeks of immersion in the mixed solution
- 5.8 Compressive strength loss of concrete after 22 weeks of immersion in the mixed solution.
- 5.9 Neutralization depth of concretes after 22 weeks of immersion in the mixed solution
- 5.10 Chloride penetration depth of concretes after 22 weeks of immersion in the mixed solution

LIST OF TABLES

Tables	Title
Chapter	2
2.1	Effect of volcanic ash and pumice powder on the compressive strength of
	cement mortar
2.2	Compressive strength of metakaolin concretes
2.3	Test result of compressive strength and elastic modulus of concrete
2.4	Water absorption for mortar with cement
2.5	Chloride permeability of blended concretes
2.6	Properties of different types of Shirasu
Chapter	3
3.1	Physical properties and chemical compositions of materials
3.2	Mixture proportions and fresh properties of the concrete specimens
Chapter	4
4.1	Mixture proportions and fresh properties of the concrete specimens
4.2	The main phases in concretes immersed in deionized water and the mixed
	solution detected by XRD method
4.3	Compressive strengths of concrete specimens after 22 weeks of immersion in
	deionized water or the mixed solution along with standard deviations
4.4	Apparent diffusion coefficients, D _C , of ions after 22 weeks of immersion in
	the mixed solution
Chapter	5
5.1	Mixture proportions and fresh properties of the concrete specimens.
5.2	The main phases in concretes immersed in deionized water and the mixed
	solution detected by XRD method

5.3 Compressive strength of concrete specimens after 22-week immersion in deionized water and the mixed solution and the standard deviation of compressive strength values.

Chapter 1. INTRODUCTION

1.1. General background

Concrete is the dominant construction material used in the infrastructure and superstructure works, therefore the improvements in the mechanical properties and durability of concrete are always desirable. However, production of cement has also caused heavy environment pressure as a result of its global contribution to CO₂ emission. The use of supplementary cementitious materials such as natural pozzolan can improves the workability, mechanical properties and durability of concrete as well as alleviates the environmental burden [1–5]. A large deposit of pyroclastic flows (the so-called Shirasu deposit) covers a wide area of approximately 3500 km2 in southern Kyushu, Japan. Previous studies [6–9] have established that Shirasu contains a high proportion of amorphous volcanic glass (approximately 60%), which exhibits pozzolanic activity. However, our knowledge of the performance of Shirasu mortar/concrete in terms of mechanical properties and durability remains limited.

Limestone is one of possible alternatives due to its wide availability and low cost. It has been reported that filler limestone powder could fill the void of cement particles and improve the packing density of cementitious materials. The incorporation of fine limestone powder can accelerates the initial hydration process, reduces the porosity of cement-based materials and improve the compressive strength concrete at early ages [10–13]. It was also found that limestone powder can be incorporated with manufactured sand to form a fine aggregate capable of filling in pores, accelerating the hydration of the cement and increasing the compressive strength and durability of the concrete [14–16].

In addition, the hydration of cement and the pozzolanic reactivity of supplementary cementitious materials such as fly ash or natural pozzolan are significantly affect by temperature [1,17–20]. It is found that the compressive strength of concrete containing supplementary cementitious materials increased remarkably under steam and autoclave curing

condition, while the flexural strength compared to those after room temperature curing. High curing temperature also resulted in decrease in porosity and improvement in strength in the interfacial transition zone (ITZ) between cement matrix and aggregate/ fiber [21–23].

In addition, degradation of concrete structure exposed to aggressive environment is a primary durability issue. Regarding the deterioration of concrete in sulfuric acid environment which can occur in concrete structures exposed to ground water, water in acid sulfate soil area [24], chemical waste or especially when deicing salt drains into a sewer system and cooperates with biogenic sulfuric acid in sewer environment [25–29]. Among ions in these environments, H^+ , SO_4^{2-} , Mg^{2+} and Cl^- ions are considered to be the most destructive ions to concrete and can deteriorate its service life and increase the maintenance cost of civil infrastructures. Therefore, there is an urgent need to study the deterioration mechanism of concrete and identify means of improving the resistance of concrete to aggressive environments. However, only a very few studies have investigated the deterioration of concretes immersed in aggressive chemicals, such as H₂SO₄ and MgCl₂ solutions that simulate the aforementioned environments. For the reasons above, the effects of limestone powder and Shirasu natural pozzolan as partial replacements for fine aggregates on the strength and resistance of concretes to an aggressive chemical environment for an aggressive chemical environment containing H₂SO₄ and MgCl₂ were examined in this study.

1.2. Objectives of the research.

The goal of this research was to provide a comprehensive understanding of the effects of Shirasu natural pozzolan and limestone powder on the strength development and the chemical resistance of concrete cured under different initial temperature through physicochemical and microstructural investigations. The specific objectives of this research were as follows:

- To study the effects of pozzolanic reactivity of Shirasu natural pozzolan and limestone powder on strength development and chemical resistance of concrete to aggressive chemical environment containing 3% sulfuric acid and 5% magnesium chloride which is produced to simulate the aggressive chemical environment.

- To investigate the effects of curing temperature on the strength and resistance of concrete to attack of the mixed solution.

- To examine the deterioration mechanism of concretes caused by the attack of the mixed solution.

1.3. Methodology of the research

First, to investigate strength development of concrete concerning the effect of limestone powder and Shirasu natural pozzolan on the hydration and pozzolanic processes; four types of concrete mixtures were produced with cylindrical specimen of size 100 mm in diameter and 200 mm in height for compressive strength measurement. The specimens were sealed and cured under different initial curing temperature conditions: one is normal curing temperature at 20 ^oC until the designated test ages; other is high curing temperature history of 60°C for 8 hours followed by cooling process. The compression test was performed at the age of 3, 28, and 91 days to examine strength development. The chemical and microstructural tests such as thermogravimetry (TG-DTA) were conducted in order to explain the strength development.

Second, to investigate the effect of limestone powder and Shirasu natural pozzolan on the resistance of concrete to the aggressive chemical environment. The immersion test in a mixed solution containing 3% sulfuric acid and 5% magnesium chloride were carried out after 56 days of cure. After 22 weeks of immersion, the erosion depth, compressive strength loss, neutralization depth, chloride penetration depth measurement were carried out to investigate the resistance of concrete to the aggressive chemical environment.

Finally, after 22 weeks of immersion, XRD analysis and the establishment of concentration profile of ions in the specimens were employed to determine the deterioration mechanism of concrete caused by the attack of the mixed solution,

3

1.4. Thesis outline

This thesis is divided into of sixes chapters, and detailed organization of the thesis is described as follows:

Chapter 1 presents the background, purposes, and methodology of this study.

Chapter 2 provides a brief literature review on strength development and durability of concrete by considering the effects of limestone powder and Shirasu natural pozzolan.

Chapter 3 presents the experimental program consisting of materials, the mixing, and casting procedure.

Chapter 4 discusses the effects of limestone powder and Shirasu natural pozzolan on the strength development and chemical resistance of concretes to the mixed solution through microstructural and chemical investigations.

Chapter 5 discusses the effects of different initial curing temperature on the strength development and chemical resistance of concrete to the mixed solution through microstructural and chemical investigations.

Chapter 6 states the conclusion of this research. Recommendations for future work are also provided.

References

[1] K. Ezziane, A. Bougara, A. Kadri, H. Khelafi, E. Kadri, Compressive strength of mortar containing natural pozzolan under various curing temperature, Cement and Concrete Composites. 29 (2007) 587–593.

[2] B. Uzal, L. Turanlı, Blended cements containing high volume of natural zeolites: Properties, hydration and paste microstructure, Cement and Concrete Composites. 34 (2012) 101–109.

[3] K. Celik, C. Meral, M. Mancio, P.K. Mehta, P.J. Monteiro, A comparative study of self-consolidating concretes incorporating high-volume natural pozzolan or high-volume fly ash, Construction and Building Materials. 67 (2014) 14–19.

[4] N. Kaid, M. Cyr, S. Julien, H. Khelafi, Durability of concrete containing a natural pozzolan as defined by a performance-based approach, Construction and Building Materials. 23 (2009) 3457–3467.

4

[5] S. Cheng, Z. Shui, T. Sun, R. Yu, G. Zhang, Durability and microstructure of coral sand concrete incorporating supplementary cementitious materials, Construction and Building Materials. 171 (2018) 44–53.

[6] K. Sodeyama, A. Tomoyose, Total utilization of Shirasu as construction materials through dry gravity classification and pulverization, Journal of Society of Material Science. 66 (2017) 574–581.

[7] K. Takewaka, T. Yamaguchi, T. Nakazaki, Long-term durability of reinforced concrete structures using "shirasu concrete" in an actual marine environment, (2008).

[8] D. Katpady, K. Takewaka, T.Yamaguchi. Experimental study on reactivity of pyroclastic flow deposit called Shirasu as binder in mortar and geopolymer. First International Conference on Concrete Sustainability, Tokyo 2013.

[9] Takewaka K., Characteristics of Shirasu concrete: based on contents in the manual established by Kagoshima prefectural office, Concrete journal. 45 (2007) 16–23.

[10] B. Lothenbach, G. Le Saout, E. Gallucci, K. Scrivener, Influence of limestone on the hydration of Portland cements, Cement and Concrete Research. 38 (2008) 848–860.

[11] P. Livesey, Strength characteristics of Portland-limestone cements, Construction and Building Materials.5 (1991) 147–150.

[12] K. De Weerdt, M.B. Haha, G. Le Saout, K.O. Kjellsen, H. Justnes, B. Lothenbach, Hydration mechanisms of ternary Portland cements containing limestone powder and fly ash, Cement and Concrete Research. 41 (2011) 279–291.

[13] A. Schöler, B. Lothenbach, F. Winnefeld, M. Zajac, Hydration of quaternary Portland cement blends containing blast-furnace slag, siliceous fly ash and limestone powder, Cement and Concrete Composites. 55 (2015) 374–382.

[14] B. Li, J. Wang, M. Zhou, Effect of limestone fines content in manufactured sand on durability of lowand high-strength concretes, Construction and Building Materials. 23 (2009) 2846–2850.

[15] H. Yang, D. Liang, Z. Deng, Y. Qin, Effect of limestone powder in manufactured sand on the hydration products and microstructure of recycled aggregate concrete, Construction and Building Materials. 188 (2018) 1045–1049.

[16] T. Çelik, K. Marar, Effects of crushed stone dust on some properties of concrete, Cement and Concrete Research. 26 (1996) 1121–1130. B. Lothenbach, F. Winnefeld, C. Alder, E. Wieland, P. Lunk, Effect of temperature on the pore solution,
 microstructure and hydration products of Portland cement pastes, Cement and Concrete Research. 37 (2007) 483–491.

[18] T. Bakharev, J.G. Sanjayan, Y.-B. Cheng, Effect of elevated temperature curing on properties of alkaliactivated slag concrete, Cement and Concrete Research. 29 (1999) 1619–1625.

[19] A.A. Ramezanianpour, M.H. Khazali, P. Vosoughi, Effect of steam curing cycles on strength and durability of SCC: A case study in precast concrete, Construction and Building Materials. 49 (2013) 807–813.

[20] Z. Wu, C. Shi, W. He, Comparative study on flexural properties of ultra-high performance concrete with supplementary cementitious materials under different curing regimes, Construction and Building Materials. 136 (2017) 307–313.

[21] H. Yazıcı, M.Y. Yardımcı, S. Aydın, A.Ş. Karabulut, Mechanical properties of reactive powder concrete containing mineral admixtures under different curing regimes, Construction and Building Materials. 23 (2009) 1223–1231.

[22] Q. Yang, S. Zhang, S. Huang, Y. He, Effect of ground quartz sand on properties of high-strength concrete in the steam-autoclaved curing, Cement and Concrete Research. 30 (2000) 1993–1998.

[23] C.-M. Aldea, F. Young, K. Wang, S.P. Shah, Effects of curing conditions on properties of concrete using slag replacement, Cement and Concrete Research. 30 (2000) 465–472.

[24] S. Mattbäck, A. Boman, P. Österholm, Hydrogeochemical impact of coarse-grained post-glacial acid sulfate soil materials, Geoderma. 308 (2017) 291–301.

[25] J. Monteny, E. Vincke, A. Beeldens, N. De Belie, L. Taerwe, D. Van Gemert, W. Verstraete, Chemical, microbiological, and in situ test methods for biogenic sulfuric acid corrosion of concrete, Cement and Concrete Research. 30 (2000) 623–634.

[26] H. Lee, R.D. Cody, A.M. Cody, P.G. Spry, Effects of various deicing chemicals on pavement concrete deterioration, in: Center for Transportation Research and Education (Ed.), Mid-Continent Transportation Symposium Proceedings, Iowa State University, Ames, USA, Citeseer, 2000: pp. 151–155.

[27] C.D. Parker, Species of sulphur bacteria associated with the corrosion of concrete, Nature. 159 (1947)439.

[28] C.D. Parker, The Corrosion of Concrete, Australian Journal of Experimental Biology and Medical Science. 23 (1945) 81–90. [29] Y. Farnam, A. Wiese, D. Bentz, J. Davis, J. Weiss, Damage development in cementitious materials exposed to magnesium chloride deicing salt, Construction and Building Materials. 93 (2015) 384–392.

Chapter 2. LITERATURE REVIEWS

2.1. Effect of limestone powder on the properties of concrete

Limestones are sedimentary rocks primarily of calcite, aragonite, vaterite and amorphous calcium carbonate. Limestone powder is crushed and ground from natural limestone or as a by-product of the limestone quarry. Limestone powder has been widely used in concrete due to its availability and low cost. It is reported that limestone powder can influence concrete properties by filler, nucleation, dilution and chemical effects. The action of limestone majorly depends on its fineness and amount [1,2].

2.1.1. Filler effect

The filler effect refines the microstructure, reduces the porosity of cement-based materials. The filler effect of limestone powder majorly relates to its particle size. If the particle size of limestone powder is finer than cement particles, limestone powder will fill the gaps between cement particles and improve the particles size distribution and finally increase the packing density of cement-based materials [3]. Although the incorporation of fine limestone powder can fill the voids between cement particle, the incorporation of limestone powder will reduce the flowability of cement-based materials if its particles size was too small since its specific surface area was high [1]. The flowability of ultrahigh-performance concrete (UHPC) decreased by 27% in the case of 3% nano limestone incorporation [4,5].

2.1.2. Nucleation effect

Nucleation effect accelerates the hydration of C_3S , increases the amount of hydration products and reduces the porosity of cement-based materials at early ages. It is reported that fine limestone powder provides nucleation sites for hydration products to deposit, accelerates the hydration reaction [6,7]. The improved precipitation of C-S-H on the surface of limestone powder was attributed to the similarity between the planar configuration of Ca and O atoms in calcite and CaO layers in C-S-H [8]. The particles size, surface structure and amount of limestone powder affect the nucleation of limestone powder [9–11]. The hydration products of cement tend to envelop the surface of limestone powder particles with similar structures which regard to Ca and O atoms in calcite and CaO layer in C-S-H. With the increase of limestone powder amount, more nucleation sites are available, more hydration products could be absorbed. The surface energy and absorption capacity of limestone powder particles increase as its particle size decreases [1]. As shown in Fig. 2.1, the main hydration peak of the sample containing the finest limestone powder (i.e., median particle size = 0.7μ m) was 25% higher than that of the OPC mixture. With the increase of particles size of limestone powder, the heat release curves become close to that of the OPC mixture. It was reported that the nucleation effect of fine limestone powder can promote the precipitation of hydration products and increase the hydration degree of cement, as a result, more hydration heat released [12].



Fig. 2.1. Influence of particles size of limestone powder (LS) on the heat release rate [9].

2.1.3. Dilution effect

The dilution effect reduces the hydration peak of C_3S and the amount of hydration products but increases the porosity of cement-based materials. The dilution effect of limestone powder on

the hydration process of cement-based materials is mainly influenced by its amount, especially when the limestone powder with coarse particle size is used. Fig. 2.2 shows that in case of incorporation of coarse limestone powder of 20 μ m, the total hydration heat of mixtures containing coarse limestone powder was lower than that of pure Portland cement mixture, and it tended to decrease with the increase of limestone powder amount [11]. It is also found that a higher replacement of cement by limestone powder remarkably reduced the Ca(OH)₂ content of cement-based materials [12].



Fig. 2.2. Heat of hydration of Portland cement-20 µm limestone [12]

2.1.4. Chemical effect

The chemical effect promotes the formation of carboaluminate and reduces the porosity of cement-based materials.

The chemical effect of limestone powder on hydration products of cement-based materials is majorly influenced by its particle size and amount and the content of alumina from C₃A and C₄AF in cement as well as its synergic effect with supplementary cementitious materials (SCMs) [1]. With the reduction in limestone powder size, the dissolution of calcium carbonate enhanced and the chemical reaction between calcium carbonate and aluminate increased [9]. The XRD data presented in Fig. 2.3 shows that the aluminate phase reacted with CaCO₃ to form mono-and hemi-carbonate when limestone powder was incorporated into cement-based materials [13]. The peak corresponding to hemicarbonate was observed after 7 days. Later, the peak of hemicarbonate decreased but the peak of monocarbonate increased. It is also found that calcium hemicarboaluminate was formed at early ages in the OPC blend containing limestone powder, and it was converted into calcium monocarboaluminate after 28 days [14]. However, it was also reported that the hydration degree of limestone powder was very low at early ages, and the calcium monocarboaluminte hydrates were detected at 180 days [15].

It is also reported that both calcium hydroxide (CH) and carboaluminate content could be influenced by the amount of limestone powder. 5% limestone powder replacement for cement decreased the normalized calcium hydroxide content after 7 days. The decrease in CH content indicated that the formation of calcium hemicarboaluminate hydrate could consume CH [13].



Fig. 2.3. XRD-patterns for the different tested blends at 1, 7, 28, 90 and 180 days. The main peaks of ettringite (E), monosulphate (M_s),possibly a sulphate and carbonate containing hydroxy-AFm(Afm*), hemicarbonate (H_c), monocarbonate (M_c) and ferrite (F) were observed [13].

2.1.5. Effect of limestone powder on fresh concrete

Workability

It is reported that the incorporation of limestone powder improved the workability of concrete [16–18]; while some scholars demonstrated that incorporating limestone powder reduced the flowability of concrete. It is also found that there was an optimum content of limestone powder for the workability of concrete [19]. Effect of limestone powder on the workability of concrete could be mainly attributed to the morphological effect, filler effect and dilution effect. The nucleation and chemical effects occur at several hours of hydration, they seem to have an insignificant effect on the workability of concrete [2].

Incorporation of limestone powder reduces the cement content, and the effective W/C increases. As a result, the electrostatic interaction between cement particles decreases, leading to the higher workability of concrete [20]. It was also reported that incorporating limestone powder reduced the water consumption of concrete [21–23] but the workability of concrete increased as the limestone powder content increased less than 20% [24,25].

Setting time

Effect of limestone powder on the setting time of concrete could mainly be attributed to the nucleation and dilution effects; while the chemical effect occurs after the setting time. When limestone powder particle size is finer than cement particles, limestone powder mainly shows the nucleation effect and dilution effect. When the particle size of limestone powder is equal to or coarser than cement particles, limestone powder mainly shows the dilution effect. Incorporating a small quantity of fine limestone powder acts as the nucleation site and reduces the setting time of concrete. The nucleation effect of limestone powder depends on its particle size and content and the mixture proportions of concrete [2]. Sato et al. [26] demonstrated that the incorporation of nano-limestone significantly accelerated hydration reactions at the early age. Acceleration of the hydration was not observed when coarser limestone powder with a median particle size of about 4 μ m [13] or 16 μ m [27] was added to replace cement. The replacement of fine aggregate with limestone powder acting as the nucleation sites resulted in

a reduction in the setting time of concrete. Due to the nucleation effect of limestone powder, the setting time of concrete increased with the decrease of the limestone powder content, as shown in Fig. 2.4.



Fig. 2.4. Setting time of concrete with 350kg/m³ of cement with different limestone powder content [28]

2.1.6. Effect of limestone powder on hardened concrete

2.1.6.1. Effect of limestone powder on shrinkage

Autogenous shrinkage of concrete occurs during the hardening of concrete due to water absorption during the hydration of cement particles. Under the drying environment, moisture loss from the surface of hardened concrete results in drying shrinkage [2]. Limestone powder is less reactive than Portland cement and other active supplementary cementitious materials, it can be considered as an inert filler material do it can reduce shrinkage of concrete [29]. It was reported that the replacement of 10% cement with limestone reduced the total shrinkage of concrete by 19% [30]. In addition, the reaction of limestone powder and C3A in cement forms calcium carboaluminate and it increases the total volume of hydrate products [31]. Therefore, the incorporation of limestone powder reduced the chemical shrinkage of concrete. In this reaction, limestone powder acted as a reactant and mitigated the chemical shrinkage of concrete [28]. The shrinkage of concrete decreased as the limestone powder content increased, as shown in Fig. 2.5 [32]. The incorporation of limestone powder reduced the cement content and increased the relative water to cement ratio, which increased the internal humidity of concrete and reduced the shrinkage of concrete [2].



Fig. 2.5. Total and autogenous shrinkages of concrete after 28 days [32]

It is reported that when limestone powder is finer than cement particles, incorporating a small amount of limestone powder to replace cement acted as the nucleation sites which enhance the hydration process. Therefore, the incorporation of limestone powder could increase the autogenous shrinkage, as shown in Fig. 2.5.

2.1.6.2. Effect of limestone powder on microstructure

Generally, the incorporation of limestone powder can modify the microstructure of cementbased materials in different manners. Fine limestone powder could fill the pores between hydration products. The nucleation effect of limestone powder could improve the hydration process and form more hydration products. The chemical effect of limestone powder can increase the volume of the solid phase of hydration products. Therefore, all of these effects influence the microstructure of cement-based materials.

It is reported that the mean pore size of mortar with plain Portland cement was 60 nm in comparison with 31nm of mean pore size of mortar containing 10% limestone powder [33]. Incorporating proper fine limestone powder content reduced the volume of large pore and

increased the volume of pores with a diameter smaller than 100nm [34,35]. In case of replacing aggregate with limestone powder, without a change in cement and water content, the threshold pore diameter of concrete containing limestone powder was lower than that of concrete without limestone powder, as shown in Fig. 2.6 [28]. In addition, the volume of large capillary pores and macro-pore of concrete containing limestone powder were less than that of concrete without limestone powder. incorporation of limestone powder to replace the aggregate filled the pores and reduced the porosity of mixtures [28].

The nucleation effect of limestone powder accelerated the hydration process, promoted the precipitation of C-S-H and filled the pores in the mixture at early ages [36,37]. In contrast, the dilution effect of limestone powder can occur when abundant coarse limestone powder was added into cement-based materials, the porosity of mixtures was higher than that of the control OPC mixture, as shown in Fig. 2.7 [38].



Fig. 2.6. Pore size distribution of mixtures with 350kg/m³ of cement [28]



Fig. 2.7. Porosity and average pore size of blended binder pastes after 28 days of hydration

[38]

2.1.6.3. Effect of limestone powder on mechanical properties

Compressive strength

Particle size, content and replacement pattern of limestone powder are factors that affect mechanical properties of concrete. Finer limestone powder could fill the void of cement particles and increase the packing density of cementitious materials. it is reported that the compressive strength of self-compacting concrete was improved because of the incorporation of ultra-fine limestone powder in concrete [39]. The compressive strength of concrete with fine limestone powder (5 μ m) was larger than that of concrete with coarse limestone powder (10 and 20 μ m) regardless of curing age [40].

When a large content of fine limestone powder replaced cement, limestone powder showed the dilution effect and resulted in the reduction of compressive strength of concrete. Generally, the compressive strength decreased as the limestone powder content increased [32,41] and the dilution effect of limestone powder was more pronounced at later age, as shown in Fig. 2.8 [42].



Fig. 2.8. Effects of limestone powder content on the compressive strength of mortars [42] When limestone powder was used to replace cement or fine aggregate, it could fill the void of cement particles and aggregate and result in the increase of compressive strength. In addition, the incorporation of limestone powder promoted the precipitation of C-S-H and increase the hydration degree of cement due to its nucleation effect. On the other hand, the increase of powder and the decrease of water might reduce the bleeding of concrete, and thus improve the bond strength of the interfacial transition zones of concrete [43,44]. However, there was an optimum content of fine limestone powder used as the replacement of cement or fine aggregate. When the content of limestone powder is less than the optimum content, finer limestone powder shows the filler effect, while finer limestone powder mainly shows the dilution effect when the amount of limestone powder surpasses the optimum content [2]. It was found that incorporating no more than 5% limestone powder to replace cement increased the early compressive strength of concrete [45–47]. Likewise, Schmidt et al. [48] found that incorporating no more than 10% limestone powder to replace cement increased the compressive strength of concrete. Similarly, Li et al. [5] reported that when 3.0% nano-CaCO₃ was incorporated into ultra-high performance concrete (UHPC) to replace cement, the compressive strength of UHPC with W/B of 0.16 and 0.17 increased by 17% and 11%, respectively.

Elastic modulus

The incorporation of limestone powder to replace cement or fine aggregate could increase the elastic modulus of concrete. The elastic modulus increased as the limestone powder content increased [43,44]. Because limestone was stiffer than the cement matrix the elastic modulus of concrete increased by 9% when the limestone powder content increased from 25% to 100% replacement rate of fine aggregate for W/C of 0.6 and 0.47, respectively [43].

When limestone powder is used to replace cement, it can decrease the elastic modulus of concrete due to the dilution effect. It is demonstrated that the incorporation of limestone powder reduced the elastic modulus when fine limestone powder is used to replace 0-45% cement. The reduction in the elastic modulus ranged between 5-33% which was proportion to the limestone content. Using less than 15% limestone powder showed little effect on the elastic modulus of concrete while using more than 15% limestone powder to replace cement significantly reduced the elastic modulus of concrete [29].

Tensile strength

Incorporation of limestone powder for replacement of cement or fine aggregate could increase the tensile strength of concrete [43,44]. The tensile strength increased with the increase of limestone content in case of using limestone to replace fine aggregate [43]. When the limestone powder content increased from 25% to 100%, the tensile strength of concrete increased by 17% and 12% for W/C of 0.6 and 0.47, respectively [43].

Flexural strength

When limestone powder is used to replace cement, it mainly shows filler and dilution effects, and the nucleation and chemical effects of limestone powder on the flexural strength of concrete are not obvious [2]. Incorporating nano-limestone with a particle size of 15–80 nm to replace cement improved the flexural strength of UHPC [5]. The flexural strength of UHPC at W/B of 0.16 and 0.17 increased by 40% and 30% respectively when 3% nano-limestone was

incorporated into UHPC to replace cement [5]. However, if the particle size of limestone was similar to cement, it would show the dilution effect. When limestone powder was used to replace cement, the content of cementing materials decreased while the non-cementing materials increased, which reduced the flexural strength of concrete. The flexural strengths reduced as the limestone powder content increased [45].

2.1.6.4. Effect of limestone powder on durability

Water permeability

It is reported that finer limestone powder can fill the voids of cement particles, reduce the porosity, and improve the water permeability of concrete [40,44,49]. Chen et al. [49] found that the water penetration depth significantly decreased as the limestone powder content increased, as shown in Fig. 2.9. It indicated that using limestone powder to replace 0–8% cement paste substantially decreased the water permeability of concrete, which could be attributed to its filler effect.



Fig. 2.9. Effects of limestone powder content on the water penetration depth of concrete [49] When a large amount of limestone powder was used to replace cement, it showed filler, nucleation and dilution effects. There could be an optimum limestone powder content for the water permeability of concrete [41].

Sorptivity

It is reported that the incorporation of limestone powder to replace cement could reduce the sorptivity coefficient of concrete [49,50]. Chen et al. [49] found that adding limestone powder to replace 0–8 wt.% cement paste effectively reduced the sorptivity coefficient of concrete due to its filler effect, as shown in Fig. 2.10.



Fig. 2.10. Effects of limestone powder content on the sorptivity of concrete [49]

Chloride permeability

When appropriate content of fine limestone powder is added to replace cement, it mainly shows the filler effect and could reduce the chloride permeability of concrete. It was found that adding limestone powder to replace cement paste also reduced the chloride diffusion coefficient [44,51], as shown in Fig. 2.11 [44]. The pore structure of concrete was refined by incorporating limestone powder due to its filler and nucleation effects, thus, the chloride diffusion coefficient reduced as the limestone powder content increased fineness, especially when the amount of limestone powder surpassed 35% as shown in Fig. 2.12 [29].



Fig. 2.11. Total charge passed versus W/C ratio in rapid chloride penetration test [44]



Fig. 2.12. Effects of different factors on the chloride diffusion coefficient of concrete [44]

Carbonation resistance

It is reported that adding limestone powder (Blaine fineness: 638 m2/kg) to replace 0–45 wt% cement significantly increased the carbonation depth of concrete, as shown in Fig. 2.13 [44]. The increase of carbonation depth might be attributed to the dilution and chemical effects of limestone powder which might consume CH in the reaction between limestone powder and aluminate phase. Parrot et al. [52] found that the natural carbonation depth increased when more

than 19% limestone powder was added to replace cement. However, the weathering carbonation degree of concrete was slight when the limestone powder content was less than 15%. Compared with concrete containing fly ash or silica fume, the carbonation depth of concrete containing limestone powder was the highest. The reason could be the larger porosity of concrete containing limestone powder [53,54].



Fig. 2.13. Carbonation depth of concrete versus limestone powder content [44]

Corrosion resistance

The electrical resistivity of concrete influences its the corrosion resistance. As the electrical resistivity of concrete increases, the movement of electrons is retarded, thereby, the corrosion resistance of concrete is improved. It was reported that the electrical resistivity decreased with the increase of limestone powder content (Blaine fineness: 330 m²/kg) [41]. Since the incorporation of 1% nano-limestone reduced capillary pores and C-S-H gels of concrete, it improved the corrosion resistance of concrete [55]. The better corrosion resistance of concrete containing limestone powder was also reported, which was accounted for the superior filler ability of limestone powder [56].
2.2. Effect of natural pozzolan on the properties of concrete

Mehta [57] classified natural pozzolan in four categories based on the identity of the pozzolanic constituents, namely siliceous and aluminous materials which in itself possess little or no cementitious properties, however in the presence of moisture will chemically react with calcium hydroxide at ordinary temperature to form cementitious compounds, as follows: unaltered volcanic glass, volcanic tuff, calcined clay or shale, and raw or calcined opaline silica.

2.2.1. Chemical and physical properties

Regarding chemical composition, many deposits contain SiO₂, Al₂O₃, CaO, Fe₂O₃, MgO, NaO, K₂O. In the most cased, the natural pozzolan is relatively high in silica compared to mineral crystals but relatively low in non- silica element (especially Mg and Fe); SiO₂ is the preponderant oxide (generally in range of 40-52% for basic lava, i.e. scoria and 63-75% for acid lava, i.e. pumice) [58–61]. Given their high silica content, natural pozzolan has a good acid resistance [62]. In a mixture of Portland cement and a pozzolan, the pozzolanic reaction progresses like an acid- base reaction between calcium hydroxide (CH) and alkalies with oxides (SiO₂+Al₂O₃+ Fe₂O₃) of pozzolan. Two phenomena occur in this mixture. First, there is a gradual decrease in the amount of free CH with time due to the consumption of CH in the pozzolanic reaction. Simultaneously, during this reaction, there is an increase of formation of C-S-H and calcium aluminosilicates and other products with low porosity that is similar to the hydration products of cement. This refinement of pores structure of cement-based materials resulted in the improvement of durability and chemical properties of concrete [57,63] The shape, fineness, particle size distribution, density, and composition of natural pozzolan particles influence the properties of fresh concrete and hardened concrete. Most of natural pozzolans tend to increase the water requirement in the normal consistency because of their microporous character and high surface area.

Chemical and physical properties of natural pozzolan can be referred with ASTM C618-19, a standard specification for fly ash and raw or calcined natural pozzolan for use as a mineral admixture in Portland cement concrete [64].

2.2.2. Reaction mechanism

The principal reaction is among amorphous SiO₂ and Al₂O₃ and CH derived from cement hydration in the presence of moisture (or water). This reaction forms additional cementitious gel C-S-H together with crystalline products, including calcium aluminate hydrates and aluminosilicate hydrates. Crystalline products depended principally on the amorphous phase of SiO₂, Al₂O₃/ CH ratio, and reaction temperature [65–67]. The pozzolanic reaction depends on the level of reactivity of natural pozzolan. Reactivity of natural pozzolan can be determined by the Chapelle test, and it expressed as a consumption rate of CH per gram of pozzolans [68–70]. Several techniques have been adopted to improve the pozzolanic reactivity of natural pozzolans, which include calcination [71–73], acid treatment [74,75], prolonged grinding [76,77], elevated temperature curing [78-80] and the use of chemical activators [81]. It has been found that the effect of calcination on the reactivity of the pozzolan depends on the combination of the two effects: an activation depending on the reactivity of the vitreous, zeolite and clay phase, and a deactivation depending on the decrease of specific surface area and soluble fraction and the increase of crystalline fraction [73]. Acid treatments result in the formation of gels on the surface of pozzolans and increase reactivity [74,75]; however, this technique is only suitable for low-calcium pozzolans [74]. Prolonged grinding of pozzolans and elevated temperature curing of concrete containing a natural pozzolan can increase reactivity, but their effects are minor. Often, concrete cured at elevated temperatures exhibits lower strengths at later ages. The addition of chemicals to concrete containing pozzolans can increase the pozzolanic reaction rate, strength development rate and ultimate strengths significantly. Although some chemicals

such as caustic alkalis are very effective, they are expensive, dangerous and not practically feasible [82].

2.2.3. Effect of natural pozzolan on fresh paste/mortar/concrete

Workability

Because of their microporous character and high surface area, typically, natural pozzolan absorbs water from the mixture and holds this water in the system allowing improved finishing of concrete [83]. However, it can result in prolonged setting time and cause a reduction in the slump of concrete [84–86]. Hossain et al. [84] reported that normal consistency of blended cement paste containing volcanic ash (VA) and pumice powder (VP) decreased by 13.8% for VA and 12.4% for VP when the VA or VP content was varied from 0 to 50%. The normal consistency decreased by 3.79% when the VP content was increased from 0 to 25% compared to 5.17% in the case of VA. The decrease in normal consistency was due to the reduction of cementitious binder in the fresh mixture with the increase of VA or VP content. On the other hand, the overall volume was increased needing more water to form a paste of the same consistency for different percentage of VA and VP in the mixture. Shannang et al. [85] reported that there was a decrease in the flow of the mortar with the increase in pozzolan content. However, in some studies [83,87] the use of natural pozzolan together with superplasticizer can enhance the workability of concrete.

Natural pozzolans generally increase the cohesiveness of the mixture by producing a more plastic paste that allows the concrete to consolidate readily and flow freely under vibration. Increased cohesiveness also helps to reduce segregation [83].

Setting time

The use of natural pozzolan can extend the setting time of concrete especially in case of replacement of cement.

Hossain et al. [84] reported that the setting time of blended cement paste increased with the increase in the replacement of cement with volcanic ash (VA) and pumice powder (VP). This was reasonable as the decrease of cement in the mixture and also decreased the surface area of the cement. As a result, the hydration process slowed down, leading to an increase in the setting time. The setting-time characteristics of concrete are influenced by ambient and concrete temperature; cement type, source, content, and fineness; water content of the paste; water-soluble alkalies; use and dosages of other admixtures; the amount of pozzolan; the fineness and chemical composition of the pozzolan [83].

2.2.4. Effect of natural pozzolan on hardened concrete

2.2.4.1. Effect of natural pozzolan on the microstructure

E.Garcia et al. [88] examined the microstructure of hydration products of blended cement. Pastes of neat and blended Portland cement (incorporating either 60% ground granulated blast furnace slag, or 30% pulverized fuel ash, or 22% volcanic ash) were cured for one year at temperatures ranging from 10°C to 60°C. Microstructure of backscattered electron images (BEI) of the OPCS–VA blended cement after curing for 1 year at 10°C and 60°C are shown in Fig. 2.14. The microstructure observed after curing at 10°C indicated a high degree of hydration of the anhydrous phases in the cement and low porosity [89], whereas the porosity after curing at 60°C was appreciably greater. The apparent porosity of the pastes increased with increasing curing temperature. Chemical analysis data for the hydration products in ternary composition diagrams indicated that in the presence of the replacement materials the composition of the C– S–H shifted towards higher Si (from 0.31 for OPC to 0.35 for OPC–VA paste) and Al contents (0.031 for OPC to 0.045 for OPC–VA paste), whereas that of Ca was lower (0.66 for OPCS to 0.61 for OPCS–VA paste).



Fig. 2.14. BEI of OPCS-VA cured for 1 year at (A) 10°C and (B) 60°C [88]

Bredy et al [90] reported that when metakaolin content was below 20%, the total porosity of the paste decreased. Khatib et al. [91] found that the incorporation of metakaolin in cement paste led to refinement of the pore structure. The threshold value for paste decreased with the increase in metakaolin content (Fig.2.15). The proportion of pores with radii smaller than 20 μ m increased with the increase in metakaolin content. Total intruded pore volumes increased between the ages of 14 and 28 days for metakaolin paste.



Fig. 2.15. Variation of threshold radius with time and metakaolin content [91]

2.2.4.2. Effect of Shirasu natural pozzolan on the mechanical properties

The effect of a natural pozzolan on the compressive strength of concrete varies markedly with the properties of the particular pozzolan and with the characteristics of the concrete mixture in which it is used.

Hossain et al. [84] studied the effect of partial replacement of cement with volcanic ash (VA) and pumice powder (VP) on the compressive strength of cement mortar. Percentage replacement varied from 0 to 50%. Tests were conducted up to the age of 28 days and results are given in Table 1. Based on the results, Hossain [84] concluded that compressive strength decreased with the increase in VA or VP content. This was due to the reduction of cement content in the mix with the increase of VA or VP content.

Table	2.1.	Effect	of	volcanic	ash	and	pumice	powder	on tl	he con	npressive	strength	of	cement
mortar	[84]]												

Mixture details	Compressive strength of mortar (MPa)									
	Volca	nic ash	(VA)		Pumice powder (VP)					
	Age (days)			Age (days)					
	1	3	7	28	1	3	7	28		
100-0	10.6	21.6	28.6	37.5	10.6	21.6	28.6	37.5		
98-2	10.0	19.9	27.1	36.0	9.9	18.6	26.4	34.5		
96-4	9.4	19.3	27.2	35.6	9.7	18.4	26.7	34.9		
94-6	9.0	18.9	26.5	34.7	9.5	18.6	26.6	34.8		
90-10	9.1	18.4	26.6	35.1	9.3	18.3	26.1	34.2		
85-15	8.7	18.1	24.2	29.6	8.9	18.0	25.6	33.2		
80-20	8.3	17.8	25.9	33.9	8.4	16.7	22.9	30.0		
75-25	7.9	17.0	22.4	26.7	8.0	15.9	22.1	28.4		
65-35	7.1	15.5	21.3	23.9	7.5	15.1	21.1	24.3		
50-50	4.2	6.4	8.1	16.5	5.1	7.4	9.2	16.4		

Brooks et al. [86] reported that compressive strength increased with the increase in the metakaolin content (Table 2). Similar results were also reported by Li et al. [92] where concrete achieved the best compressive strength with 10% metakaolin content. Poon et al. [93] demonstrated that cement pastes containing 5% to 20% metakaolin had higher compressive strengths than the control at all ages from 3 to 90 days, with the paste containing 10% metakaolin performing the best. Jin and Li [94] concluded that the incorporation of metakaolin can improve both the strength and modulus of elasticity development of a young concrete, and metakaolin showed the best enhancement on the mechanical properties of young concrete.

Table 2.2. Compressive strength of metakaolin concretes [86]

Concrete mixes	Compressive strength (MPa)
OPC	87.0
MK5	91.5
MK10	104.0
MK15	103.5

Badogiannis et al. [95] studied the compressive strength of cement containing five metakaolins up to 180 days. The metakaolinite contents in metakaolins MK1, MK2, MK3, and MK4 (derived from poor Greek kaolins) were 36, 37, 71, and 49%, respectively, but 95% in a commercial metakaolin (MKC) of high purity. Metakaolin had a very positive effect on the cement strength after 2 days and specifically at 28 and 180 days. 10% of metakaolin content seemed to be, generally, more favorable than 20%.

Kaid et al. [96] reported that concrete using natural pozzolan as replacement of binder and fine aggregate exhibited improvement of mechanical properties of concrete after 28 days of cure

	Curing time	RefPz	RefCem	ConPz
Compressive strength	28	31.2 ± 2.4	33.3 ± 0.9	36.4 ± 2.4
(MPa)	60	41.0 ± 1.9	44.5 ± 0.8	41.8 ± 1.7
	180	43.9 ± 1.6	45.7 ± 1.9	44.9 ± 0.6

24.6

 44.4 ± 0.6

 46.0 ± 1.9

24.0

Table 2.3. Test result of compressive strength and elastic modulus of concrete [96]

RefPz: Concrete with a binder composed of 80% CEM I and 20% NP was chosen as the reference Algerian concrete.

RefCem: Pozzolan-free concrete made of pure cement (CEM I)

270

60

ConPz: This concrete was designed to improve the durability of concretes with NP. It had the same binder as RefPz (80% CEM I – 20% NP), but with a supplementary quantity of NP added as sand replacement (13% of the mass of fine aggregate 0/0.315 mm)

2.2.4.3. Effect of Shirasu natural pozzolan on the durability

Alkali-silica reaction

Elastic modulus (GPa)

Ramlochan et al. [97] reported that the incorporation of high-reactivity metakaolin (HRM) as a partial cement replacement between 10 and 15% may be sufficient to control deleterious expansion due to alkali-silica reaction in concrete, depending on the nature of the aggregate. The mechanism by which HRM may suppress expansion due to alkali-silica reaction appeared to be entrapment of alkalis by the supplementary hydrates and a consequent decrease in the pH of pore solutions.

 45.1 ± 0.5

24.6

Water absorption and sorptivity

Courard et al. [98] indicated that water absorption of concrete mixtures increased with the increase in natural pozzolan content at 28 days and 14 months (Table 2.4). Khatib et al. [99] reported the increase in water absorption (WA) of concrete mixtures with the increase in natural pozzolan contents at all curing times (Fig. 2.16).

Material	Water absorption (% mass)				
	After 28 days	After 14 months			
CEM I 42.5	8.16	7.82			
5% metakaolin	8.39	8.04			
10% metakaolin	8.78	8.44			
15% metakaolin	9.71	8.77			
20% metakaolin	9.70	8.97			
10% kaolin	9.51	7.90			

Table 2.4. Water absorption for mortar with cement CEM I, metakaolin and kaolin [98]



Fig. 2.16. Variation of water absorption of concrete containing metakaolin [99]

Chloride permeability

It is reported that concrete containing metakaolin showed significantly lower conductivity values than the Portland cement concrete [100]. Hooton et al [101] showed that 8 and 12% by mass of highly reactive metakaolin improved the chloride penetration resistance of both 0.30

and 0.40 w/cm concretes. Bai et al. [102] reported that significant reductions in chloride penetration depths were observed in the case of a partially replacement of cement with metakaolin in concrete exposed to seawater.

Poon et al. [103] investigated the chloride permeability of the concrete with metakaolin at w/b ratios of 0.3 and 0.5 were reported by (Table 2.5). Both the metakaolin concretes showed lower total ion penetration than the control concrete without any replacement of cement with mineral admixture. At w/b of 0.3, concrete with a 10% metakaolin showed the best performance, while at w/b of 0.5, 20% replacement was the best.

Series	w/b ratio	Mix	Tota	l charge pa	ssed (Could	ombs)
			3 days	7 days	28 days	90 days
1	0.30	Control	2461	2151	1035	931
		5% MK	1327	1244	862	646
		10% MK	417	347	199	135
		20% MK	406	395	240	124
2	0.50	Control	5312	4054	2971	2789
		5% MK	4215	3765	2079	1065
		10% MK	1580	1247	918	752
		20% MK	751	740	640	580

 Table 2.5. Chloride permeability of blended concretes [103]

Corrosion resistance, electrical resistivity

The use of metakaolin, either as a sand replacement up to 20% or as a cement replacement up to 10%, improved the corrosion behavior of mortar specimens, while there is an insignificant effect in case of metakaolin added in greater percentages [104].

The electrical resistivity is an indirect measurement of porosity and diffusivity. Hossain et al. [105] reported the results of the electrical resistivity of volcanic ash blended cement mortar. The mortars had 20% and 40% volcanic ash (VA) as cement replacement and water/binder ratio of 0.55. They pointed out that the electrical resistivity of the cement–VA mortars increased rapidly with the curing time. The 40%VA mortars had shown substantially higher resistivity than 20% VA mortars whereas the resistivity of the control mortar (0% VA) did not increase

with the curing time especially in the long term and remained lower than the VA mortars. Electrical current through hydrating cement mortar is electrolytic (i.e., mainly due to the flow of ions through the pore spaces). It would enhance the overall resistivity of VA blended concrete and hence a lower degree of bar corrosion [105].



Fig. 2.17. Variation of electrical resistivity of concrete containing volcanic ash [105]

2.2.5. Shirasu natural pozzolan

Shirasu, a kind of the volcanic ash, which was abundantly deposited because of a huge pyroclastic flow covers a wide area of approximately 3500 km² in the southern part of Kyushu island, Japan, in 20 to 100 thousand years ago. Shirasu is sandy but porous material with a large amount of very fine particles. Fig. 2.19 showes microscopic of Shirasu which confirms the porous nature of Shirasu. It has about 80% density of that of sea sand and three times higher water absorption capacity when compared to sea sand. It also has a high quantity of volcanic glasses in its mineral compositions and has been confirmed to show the pozzolanic reactivity. The quantity of volcanic glasses in Shirasu's composition is approximately 80% [106]. Therefore, it should be a potential construction material used as the replacement of binder, fine aggregate in mortar, concrete and geopolymer [107]. It is reported that Shirasu concrete has

shown long-term durability concerning salt attack, sulfate attack, alkali-aggregate reaction, and etc. [108].



Fig. 2.18. Source of Shirasu [108]



Fig. 2.19. Microscopic of Shirasu particle [108]

The different types of Shirasu deposits available in Kagoshima Prefecture of Japan at different locations based on density current and origin are shown in Table 2.6.

Table 2.6. Properties of different types of Shirasu [109].

	Density of	Absorption		
Type of	surfaça dry stata	conscitu ratio	Fineness	Fine powder
	surface-ury state	capacity ratio	modulus	ratio (%)
Shirasu/location	(g/cm^3)	(%)		
Kagoshima shi	2.06	11.31	1.53	21.9
Yoshida	2.22	4.29	1.57	20.54
Kiire	2.16	4.2	1.68	16.64
Makurazaki	2.24	3.11	1.18	23.4
Kaseda	2.16	8.48	1.12	27.55
Kawanabe	2.24	2.51	1.26	24.44
Higashi Ichiki	2.24	3.6	1.56	20.02
Koriyama	2.22	6.03	1.73	20.6
Ijuin	2.24	3.23	1.62	20.37
Matsumoto	2.22	3.07	1.63	24.46
Hioshi	2.24	2.67	1.6	21.3
Fukiage	2.19	4.29	1.43	26.63
Kimpo	2.22	3.07	1.14	19.48
Hiwaki	2.27	3.83	1.69	16.35
Tsuruta	2.25	3.61	1.44	26.41
Satsuma	2.23	3.82	1.52	24.06
Iriki	2.3	2.6	1.57	18.64
Izumi	2.22	5.07	1.21	25.52
Mizobe	2.16	6.4	1.83	17.91
Yokogawa	2.2	6.54	1.57	20.11
Kirishima	2.19	4.91	1.68	20.79
Fukuyama	2.11	9.62	2.09	20.86
Tarumizu	2.18	5.46	1.43	24.89
Takarabe	2.18	4.77	1.68	21.49
Shibushi	2.2	4.02	1.27	27.36
Sueyoshi	2.21	4.28	1.51	25.04
Matsuyama	2.24	3.11	1.5	24.53
Kihoku	2.17	3.62	1.87	22.35
Ariake	2.19	6.72	1.47	25.45
Takayama	2.19	4.17	1.5	23.36
Aira	2.17	2.88	1.37	24.8

2.3. Effect of curing temperature on properties of concrete

Many studies reported the effect of temperature on the pore solution, microstructure and hydration products of Portland cement pastes and pozzolanic activity of supplementary cementitious materials [110–114]. Based on a review of the references it is evident that the curing temperature significantly influences the mechanical properties and durability of concrete. Some experimental results [110,115] indicated that increasing the early age curing temperature

tends to increase the compressive strength of concrete. This could be explained by the initial elevated temperature curing, which promoted the hydration reaction of cement and the pozzolanic reaction of the mineral admixture. However, some studies [114–116] demonstrated that at later-age the mechanical properties and durability of concrete are degraded with the increase of the early curing temperatures. Because the rapid hydration does not allow for sufficient time for the hydration products to diffuse within the voids and retards its further hydration. Therefore, the calcium silicate hydrates distribute non-uniformly and lead to a formation of large pores and a reduction in the strength of concrete. These behaviors had been called the crossover effect [117].

2.4. Deterioration of concrete in aggressive environment

2.4.1. Deterioration of concrete caused by acid environment

The aggressive acidic environment was derived from a variety of natural sources and industrial processes. In cases of free acids in industrial processes, the concentration of acid contacting with concrete structures may reach high values. The source of acid media can even represent air pollution by gaseous carbon dioxide, sulfur dioxide and nitrogen oxides. The acidic attack may also occur as a consequence of bacterial activity, leading to the formation of sulfuric acid in groundwater, sewage system or in the sulfate soil areas, etc. [118–121]. The acidic attack of cement-based materials is based on their alkalinity. If concrete structures are attacked by an acid, the components of the cement matrix, including calcium hydroxide and other hydration products, break down in accordance with the acid-base reaction. They are complex series of processes of decomposition and leaching of the constituents of the cement matrix. The decomposition and leaching of the acid, the solubility of the acid calcium salts reaction products and on the fluid transport through the concrete (static condition or dynamic condition with change of level or flow of the acid solution)

[122,123]. Insoluble calcium salts may precipitate in the voids and can slow down the attack. Acids such as nitric acid, hydrochloric acid and acetic acid are very aggressive as their calcium salts are readily soluble and removed from the attack front. Other acids, such as phosphoric acid, are less harmful as their calcium salt, due to their low solubility, inhibits the attack by blocking the pathways within the concrete such as interconnected cracks, voids and porosity. Sulfuric acid is very damaging to concrete as it combines an acid attack and a sulfate attack [124]. The deterioration of concrete by the sulfuric acid attack can generally be characterized by equations (1)-(5), as follows [125] :

$$Ca(OH)_2 + H_2SO_4 \rightarrow CaSO_4.2H_2O \tag{2.1}$$

$$C_{3}S_{2}H_{3} + 3H_{2}SO_{4} + 4H_{2}O \rightarrow 3(CaSO_{4}.2H_{2}O) + 2(SiO_{2}.2H_{2}O)$$
(2.2)

$$C_3AH_6 + 6H_2SO_4 \rightarrow 3(CaSO_4.2H_2O) + Al_2(SO_4)_3 + 6H_2O$$
 (2.3)

$$C_{6}AF_{2}H_{12} + 11 H_{2}SO_{4} \rightarrow 6(CaSO_{4}.2H_{2}O) + Al_{2}(SO_{4})_{3} + 2FeSO_{4} + 11H_{2}O$$
(2.4)

$$C_{3}AH_{6} + 3(CaSO_{4.}2H_{2}O) + 14 H_{2}O \rightarrow 3CaO_{4.}Al_{2}O_{3.}3 CaSO_{4.}32H_{2}O$$
 (2.5)

The reaction products formed on the concrete surface are gypsum and possibly ettringite which are associated with the volume expansion, leading to tensile stress in concrete, cracking and spalling. It indicates that sulfuric acid attacked the cement matrix by decalcifying calcium hydroxide and hydration products and generating expansive reaction products, thus resulting in strength loss and crack occurrence in the concrete structure.

2.4.2. Deterioration of concrete caused by magnesium chloride

Magnesium chloride (MgCl₂) is commonly used in deicing applications due to its capability to depress freezing temperatures to a lower point than other salts such as sodium chloride (NaCl). However, it is reported that concrete exposed to MgCl₂ deicing salt typically exhibits changes in its microstructure due to chemical reactions, including formation of brucite Mg(OH)₂, Friedel's salts (3Ca(OH)₂. CaCl₂. 10H₂O), magnesium silicate hydrate (M–S–H), magnesium oxychloride (Mg(OH)₂.MgCl₂.8 H₂O), and/or secondary calcium oxychloride (3Ca(OH)₂. CaCl₂. 12H₂O); these changes can result in severe cracking, even if the concrete does not experience any freezing and thawing cycles [126–129], as following equations:

$$C-S-H + MgCl_2 \rightarrow CaCl_2 + M-S-H$$
(2.6)

$$Ca(OH)_2 + CaCl_2 + 12H_2O \rightarrow 3Ca(OH)_2. CaCl_2. 12H_2O$$

$$(2.7)$$

$$MgCl_2 + 3Ca(OH)_2. CaCl_2. 12H_2O \rightarrow 3Ca(OH)_2. CaCl_2. 10H_2O + MgSO_{4.2} H_2O \qquad (2.8)$$

$$Ca(OH)_2 + MgCl_2 \rightarrow CaCl_2 + Mg(OH)_2$$
(2.9)

$$(3 \text{ or } 5)Mg(OH)_2 + MgCl_2 + 8 H2O \rightarrow Mg(OH)_2.MgCl_2.8 H2O$$

$$(2.10)$$

It is reported that non-cementitious magnesium silicate hydrate (M-S-H) is formed by replacing the calcium from the cementitious calcium silicate hydrate (C–S–H), leading to the disintegration of cement matrix [130,131]. While the formation of calcium oxychloride has been reported to be very expansive and destructive within the cementitious matrix [127,129]. In contrast, the formations of, brucite, magnesium sulfate, and Freidel's salt are not generally reported as very destructive components.

2.4.3. Some methods of improving the resistance of concrete to aggressive environments

2.4.3.1. Type of cement

The type of cement used has also an important bearing on the performance of concrete in an aggressive environment [132]. It seems that high-alumina cement maybe show a high resistant to acidic attack solutions with pH values between 4 and 5. The better performance of these cement compared to that of Portland cement could be attributed not only to the absence of calcium hydroxide but to the presence of more stable calcium-aluminate hydrates in the cement matrix and also to the presence of aluminum hydroxide. This may encapsulate the hydration products and protect them from the acidic attack [133].

2.4.3.2. The incorporation of mineral admixture

The incorporation of mineral admixture is a common means of improving the resistance of concrete to the aggressive environment. Contradictory results have been reported with the

replacement of cement with silica fume [134,135], blast furnace slag [136,137], fly ash [138,139], natural pozzolan [33,140–142]. Makhloufi et al. [143] showed that the incorporation of limestone, natural pozzolan, blast furnace slag or silica fume improves the mechanical properties and sulfuric acid resistance of blended cement mortars. The addition of natural pozzolan also improves resistance to sulfuric acid and hydrochloric acid and reduces the penetration of sulfate and chloride ions [144–146].

2.4.3.3. Water-cement ratio

It is known that the value of w/c ratio has a significant influence on cement hydration, microstructure and porosity of concrete structure. It is also known that the quality of concrete is increased with the w/c ratio decrease. The high w/c ratio used for composite mixtures will exhibit lower values of strength, bulk volume and higher porosity and potentially lower resistance to the aggressive environment [147,148]. Since one of the most important factors related to the mechanism of all the deterioration phenomena is permeability of concrete, while permeability of concrete is significantly influenced by w/c ratio. It is reported that the permeability is significantly reduced to a w/c ratio below 0.45. Therefore, in the interests of ensuring of an acid-resistant concrete, it is necessary that the w/c ratio should be less than 0.45 [132].

2.5. Summary

The incorporation of mineral admixture is a common means of improving the mechanical properties and durability of concrete. In addition, concrete structures must withstand a variety of environmental stresses. Deicing salt, saline, acid sulfate soils and industrial drainage can all degrade the strength of concrete. In these environments, H⁺, SO₄²⁻, Mg²⁺, Cl⁻ ions are considered as the most destructive ions for concrete structures. The deterioration of concrete caused by the attack of the mixed solution containing H⁺, SO₄²⁻, Mg²⁺, Cl⁻ ions is a complex mechanism related to a series of physiochemical factors. However, very limited studies have examined the

detrimental effect of these ions on the durability of concrete which associated with the service life and maintenance cost of civil infrastructure. Therefore, there is an urgent need to study the deterioration mechanism of concrete and identify means of improving the resistance of concrete to this aggressive environment.

For the aforementioned reasons, the effects of limestone powder and Shirasu natural pozzolan, abundant volcanic ash in Kyushu island of Japan, as fine aggregate replacements in concrete were investigated in this study with respect to the compressive strength development and the resistance of concrete to aggressive environment composing of sulfuric acid and magnesium chloride. In addition, the influence of initial curing temperature on the mechanical properties and the durability of concrete were investigated in order to determine the most effective curing conditions for concretes regarding technical and economic considerations.

References in chapter 2

- D. Wang, C. Shi, N. Farzadnia, Z. Shi, H. Jia, Z. Ou, A review on use of limestone powder in cement-based materials: Mechanism, hydration and microstructures, Construction and Building Materials. 181 (2018) 659–672.
- [2] D. Wang, C. Shi, N. Farzadnia, Z. Shi, H. Jia, A review on effects of limestone powder on the properties of concrete, Construction and Building Materials. 192 (2018) 153–166.
- [3] EJ Sellevold, DH Bager, EK Jensen, T Knudsen, Silica fume cement pastes: hydration and pore structure, in: condensed silica fume in concrete, Institute for Bygningsmaterialae, Norges Tekniske, Norway, BML 1982, 82.610...
- [4] W. Li, Z. Huang, T. Zu, C. Shi, W.H. Duan, S.P. Shah, Influence of nanolimestone on the hydration, mechanical strength, and autogenous shrinkage of ultrahigh-performance concrete, Journal of Materials in Civil Engineering. 28 (2015) 04015068.
- [5] W. Li, Z. Huang, F. Cao, Z. Sun, S.P. Shah, Effects of nano-silica and nano-limestone on flowability and mechanical properties of ultra-high-performance concrete matrix, Construction and Building Materials. 95 (2015) 366–374.
- [6] I. Soroka, N. Stern, Calcareous fillers and the compressive strength of Portland cement, Cement and Concrete Research. 6 (1976) 367–376.

- [7] B. Craeye, G. De Schutter, B. Desmet, J. Vantomme, G. Heirman, L. Vandewalle, Ö. Cizer, S. Aggoun, E.H. Kadri, Effect of mineral filler type on autogenous shrinkage of self-compacting concrete, Cement and Concrete Research. 40 (2010) 908–913.
- [8] S. Rode, N. Oyabu, K. Kobayashi, H. Yamada, A. Kühnle, True atomic-resolution imaging of (1014) calcite in aqueous solution by frequency modulation atomic force microscopy, Langmuir. 25 (2009) 2850–2853.
- [9] K. Vance, M. Aguayo, T. Oey, G. Sant, N. Neithalath, Hydration and strength development in ternary portland cement blends containing limestone and fly ash or metakaolin, Cement and Concrete Composites. 39 (2013) 93–103.
- [10] D.P. Bentz, A. Ardani, T. Barrett, S.Z. Jones, D. Lootens, M.A. Peltz, T. Sato, P.E. Stutzman, J. Tanesi,
 W.J. Weiss, Multi-scale investigation of the performance of limestone in concrete, Construction and
 Building Materials. 75 (2015) 1–10.
- [11] P. Thongsanitgarn, W. Wongkeo, A. Chaipanich, C.S. Poon, Heat of hydration of Portland high-calcium fly ash cement incorporating limestone powder: Effect of limestone particle size, Construction and Building Materials. 66 (2014) 410–417.
- [12] J. Tikkanen, A. Cwirzen, V. Penttala, Effects of mineral powders on hydration process and hydration products in normal strength concrete, Construction and Building Materials. 72 (2014) 7–14.
- [13] K. De Weerdt, M.B. Haha, G. Le Saout, K.O. Kjellsen, H. Justnes, B. Lothenbach, Hydration mechanisms of ternary Portland cements containing limestone powder and fly ash, Cement and Concrete Research. 41 (2011) 279–291.
- [14] A. Ipavec, R. Gabrovšek, T. Vuk, V. Kaučič, J. Maček, A. Meden, Carboaluminate Phases Formation During the Hydration of Calcite-Containing Portland Cement, Journal of the American Ceramic Society. 94 (2011) 1238–1242.
- [15] S. Liu, P. Yan, Hydration properties of limestone powder in complex binding material, Journal of the Chinese Ceramic Society. 36 (2008) 1401–1405.
- [16] G. Sua-iam, N. Makul, Use of limestone powder during incorporation of Pb-containing cathode ray tube waste in self-compacting concrete, Journal of Environmental Management. 128 (2013) 931–940.
- [17] R. Derabla, M.L. Benmalek, Characterization of heat-treated self-compacting concrete containing mineral admixtures at early age and in the long term, Construction and Building Materials. 66 (2014) 787–794.
- [18] K. Kanazawa, K. Yamada, S. Sogo, Properties of low-heat generating concrete containing large volumes of blast-furnace slag and fly ash, Special Publication. 132 (1992) 97–118.

- [19] A. Yahia, M. Tanimura, Y. Shimoyama, Rheological properties of highly flowable mortar containing limestone filler-effect of powder content and W/C ratio, Cement and Concrete Research. 35 (2005) 532– 539.
- [20] K. Vance, A. Kumar, G. Sant, N. Neithalath, The rheological properties of ternary binders containing Portland cement, limestone, and metakaolin or fly ash, Cement and Concrete Research. 52 (2013) 196–207.
- [21] E.F. Irassar, D. Violini, V.F. Rahhal, C. Milanesi, M.A. Trezza, V.L. Bonavetti, Influence of limestone content, gypsum content and fineness on early age properties of Portland limestone cement produced by inter-grinding, Cement and Concrete Composites. 33 (2011) 192–200.
- [22] O. Burgos-Montes, M.M. Alonso, F. Puertas, Viscosity and water demand of limestone- and fly ash-blended cement pastes in the presence of superplasticisers, Construction and Building Materials. 48 (2013) 417–423. https://doi.org/10.1016/j.conbuildmat.2013.07.008.
- [23] S. Tsivilis, E. Chaniotakis, G. Kakali, G. Batis, An analysis of the properties of Portland limestone cements and concrete, Cement and Concrete Composites. 24 (2002) 371–378.
- [24] M. Schmidt, K. Harr, R. Boeing, Blended cement according to ENV 197 and experiences in Germany, Cement, Concrete and Aggregates. 15 (1993) 156–164.
- [25] G.K. Moir, S. Kelham, Developments in the manufacture and use of Portland limestone cement, Special Publication. 172 (1999) 797–820.
- [26] T. Sato, J.J. Beaudoin, Effect of nano-CaCO3 on hydration of cement containing supplementary cementitious materials, Advances in Cement Research. 23 (2011) 33–43.
- [27] D.P. Bentz, Powder additions to mitigate retardation in high-volume fly ash mixtures, ACI Materials Journal. 107 (2010) 508.
- [28] M. Valcuende, C. Parra, E. Marco, A. Garrido, E. Martínez, J. Cánoves, Influence of limestone filler and viscosity-modifying admixture on the porous structure of self-compacting concrete, Construction and Building Materials. 28 (2012) 122–128.
- [29] M.S. Meddah, M.C. Lmbachiya, R.K. Dhir, Potential use of binary and composite limestone cements in concrete production, Construction and Building Materials. 58 (2014) 193–205.
- [30] S. Tongaroonsri, S. Tangtermsirikul, Effect of mineral admixtures and curing periods on shrinkage and cracking age under restrained condition, Construction and Building Materials. 23 (2009) 1050–1056.
- [31] W. Chong, W. Yongwei, P.U. Xincheng, Characteristics and Mechanisms for Autogenous Shrinkage of Cement Concrete with Very Low Water-binder Ratio, Journal of Building Materials. 13 (2010) 75–79.

- [32] L. Courard, F. Michel, Limestone fillers cement based composites: effects of blast furnace slags on fresh and hardened properties, Construction and Building Materials. 51 (2014) 439–445.
- [33] Y. Senhadji, G. Escadeillas, M. Mouli, H. Khelafi, Influence of natural pozzolan, silica fume and limestone fine on strength, acid resistance and microstructure of mortar, Powder Technology. 254 (2014) 314–323.
- [34] S.-H. Liu, J. Wang, Influence of limestone powder on pore structure of mortar, Jianzhu Cailiao Xuebao/Journal of Building Materials. 14 (2011) 532-535+575. https://doi.org/10.3969/j.issn.1007-9629.2011.04.019.
- [35] W. Kai, S. Huisheng, X. Linglin, Effect of mineral admixture on mechanical properties of concrete by adjusting interfacial transition zone microstructure, Journal of the Chinese Ceramic Society. 45 (2017) 623– 630.
- [36] R. Zaitri, M. Bederina, T. Bouziani, Z. Makhloufi, M. Hadjoudja, Development of high performances concrete based on the addition of grinded dune sand and limestone rock using the mixture design modelling approach, Construction and Building Materials. 60 (2014) 8–16.
- [37] T. Sato, F. Diallo, Seeding effect of nano-CaCO3 on the hydration of tricalcium silicate, Transportation Research Record. 2141 (2010) 61–67.
- [38] S. Das, M. Aguayo, V. Dey, R. Kachala, B. Mobasher, G. Sant, N. Neithalath, The fracture response of blended formulations containing limestone powder: Evaluations using two-parameter fracture model and digital image correlation, Cement and Concrete Composites. 53 (2014) 316–326.
- [39] V.B. Bosiljkov, SCC mixes with poorly graded aggregate and high volume of limestone filler, Cement and Concrete Research. 33 (2003) 1279–1286.
- [40] N. Ghafoori, R. Spitek, M. Najimi, Influence of limestone size and content on transport properties of selfconsolidating concrete, Construction and Building Materials. 127 (2016) 588–595.
- [41] A.A. Ramezanianpour, E. Ghiasvand, I. Nickseresht, M. Mahdikhani, F. Moodi, Influence of various amounts of limestone powder on performance of Portland limestone cement concretes, Cement and Concrete Composites. 31 (2009) 715–720.
- [42] G.İ. Sezer, Compressive strength and sulfate resistance of limestone and/or silica fume mortars, Construction and Building Materials. 26 (2012) 613–618.
- [43] I.M. Nikbin, M.H.A. Beygi, M.T. Kazemi, J.V. Amiri, S. Rabbanifar, E. Rahmani, S. Rahimi, A comprehensive investigation into the effect of water to cement ratio and powder content on mechanical properties of self-compacting concrete, Construction and Building Materials. 57 (2014) 69–80.

- [44] L.G. Li, A.K. Kwan, Adding limestone fines as cementitious paste replacement to improve tensile strength, stiffness and durability of concrete, Cement and Concrete Composites. 60 (2015) 17–24.
- [45] P. Pliya, D. Cree, Limestone derived eggshell powder as a replacement in Portland cement mortar, Construction and Building Materials. 95 (2015) 1–9.
- [46] P. Livesey, Strength characteristics of Portland-limestone cements, Construction and Building Materials. 5 (1991) 147–150.
- [47] A. Schöler, B. Lothenbach, F. Winnefeld, M. Zajac, Hydration of quaternary Portland cement blends containing blast-furnace slag, siliceous fly ash and limestone powder, Cement and Concrete Composites. 55 (2015) 374–382.
- [48] M. Schmidt, Cement with interground additives-capabilities and environmental relief: I, ZKG International, Edition B. 45 (1992) 64–9.
- [49] J.J. Chen, A.K.H. Kwan, Y. Jiang, Adding limestone fines as cement paste replacement to reduce water permeability and sorptivity of concrete, Construction and Building Materials. 56 (2014) 87–93.
- [50] M. Gesoğlu, E. Güneyisi, M.E. Kocabağ, V. Bayram, K. Mermerdaş, Fresh and hardened characteristics of self compacting concretes made with combined use of marble powder, limestone filler, and fly ash, Construction and Building Materials. 37 (2012) 160–170.
- [51] H. Hornain, J. Marchand, V. Duhot, M. Moranville-Regourd, Diffusion of chloride ions in limestone filler blended cement pastes and mortars, Cement and Concrete Research. 25 (1995) 1667–1678.
- [52] L.J. Parrott, Some effects of cement and curing upon carbonation and reinforcement corrosion in concrete, Materials and Structures. 29 (1996) 164.
- [53] M.K. Mohammed, A.R. Dawson, N.H. Thom, Carbonation of filler typed self-compacting concrete and its impact on the microstructure by utilization of 100% CO2 accelerating techniques, Construction and Building Materials. 50 (2014) 508–516.
- [54] Z. Zhang, Q. Wang, H. Chen, Properties of high-volume limestone powder concrete under standard curing and steam-curing conditions, Powder Technology. 301 (2016) 16–25.
- [55] F.U.A. Shaikh, S.W. Supit, Chloride induced corrosion durability of high volume fly ash concretes containing nano particles, Construction and Building Materials. 99 (2015) 208–225.
- [56] S. Ahmad, S.K. Adekunle, M. Maslehuddin, A.K. Azad, Properties of self-consolidating concrete made utilizing alternative mineral fillers, Construction and Building Materials. 68 (2014) 268–276.

- [57] P.K.Mehta, Natural Pozzolans: Supplementary Cementing Materials for Concrete, (1987) CANMET-SP 86-8E, Canadian Government Publishing Center, Supply and Services, Ottawa, Canada, K1A0S9...
- [58] R. Siddique, Effect of volcanic ash on the properties of cement paste and mortar, Resources, Conservation and Recycling. 56 (2011) 66–70.
- [59] C. Leonelli, E. Kamseu, D.N. Boccaccini, U.C. Melo, A. Rizzuti, N. Billong, P. Miselli, Volcanic ash as alternative raw materials for traditional vitrified ceramic products, Advances in Applied Ceramics. 106 (2007) 135–141.
- [60] K.M.A. Hossain, Volcanic ash and pumice as cement additives: pozzolanic, alkali-silica reaction and autoclave expansion characteristics, Cement and Concrete Research. 35 (2005) 1141–1144.
- [61] S. Demirdag, I. Ugur, S. Sarac, The effects of cement/fly ash ratios on the volcanic slag aggregate lightweight concrete masonry units, Construction and Building Materials. 22 (2008) 1730–1735.
- [62] P.N. Lemougna, K. Wang, Q. Tang, A.N. Nzeukou, N. Billong, U.C. Melo, X. Cui, Review on the use of volcanic ashes for engineering applications, Resources, Conservation and Recycling. 137 (2018) 177–190.
- [63] A.M. Neville, Properties of Concrete. (5th edition), Addison Wesley, Longman, England (1996)...
- [64] ASTM C618-19, Standard Specification for Coal Fly Ash and Raw or Calcined Natural Pozzolan for Use in Concrete...
- [65] M. Murat, Hydration reaction and hardening of calcined clays and related minerals. I. Preliminary investigation on metakaolinite, Cement and Concrete Research. 13 (1983) 259–266.
- [66] S. Wild, J.M. Khatib, L.J. Roose, Chemical shrinkage and autogenous shrinkage of Portland cement metakaolin pastes, Advances in Cement Research. 10 (1998) 109–119.
- [67] J.M. Kinuthia, S. Wild, B.B. Sabir, J. Bai, Self-compensating autogenous shrinkage in Portland cement metakaolin—fly ash pastes, Advances in Cement Research. 12 (2000) 35–43.
- [68] T. Perraki, G. Kakali, E. Kontori, Characterization and pozzolanic activity of thermally treated zeolite, Journal of Thermal Analysis and Calorimetry. 82 (2005) 109–113.
- [69] G.C. Cordeiro, R.D. Toledo Filho, L.M. Tavares, E.M.R. Fairbairn, Pozzolanic activity and filler effect of sugar cane bagasse ash in Portland cement and lime mortars, Cement and Concrete Composites. 30 (2008) 410–418.
- [70] G. Kakali, T.H. Perraki, S. Tsivilis, E. Badogiannis, Thermal treatment of kaolin: the effect of mineralogy on the pozzolanic activity, Applied Clay Science. 20 (2001) 73–80.

- [71] R.C. Mielenz, L.P. Witte, O.J. Glantz, Effect of calcination on natural pozzolans, in: Symposium on Use of Pozzolanic Materials in Mortars and Concretes, ASTM International, 1950.
- [72] G. Rossi, L. Forchielli, Porous structure and reactivity with lime of some natural Italian pozzolans, II Cemento. 73 (1976) 215–221.
- [73] U. Costa, F. Massazza, Influence of thermal treatment on the activity of some natural pozzolans with lime, Cemento. 74 (1977) 105–122.
- [74] R.T. Hemmings, E.E. Berry, B.J. Cornelius, D.M. Golden, Evaluation of acid-leached fly ash as a pozzolan, MRS Online Proceedings Library Archive. 136 (1988).
- [75] S. Teichner, E. Pernoux, The texture and surface area of kieselguhrs after various treatments, Clay Minerals Bulletin. 1 (1951) 145–150.
- [76] K.M. Alexander, Reactivity of ultrafine powders produced from siliceous rocks, in: Journal Proceedings, 1960: pp. 557–570.
- [77] M.K. Chatterjee, D. Lahiri, Pozzolanic Activity in Relation to Specific Surface of Some Artificial Pozzolanas—Part I: Measurement of Surface Area from Water Adsorption Isotherms, Transactions of the Indian Ceramic Society. 26 (1967) 65–70.
- [78] K.M. Alexander, J. Wardlaw, Limitations of the pozzolana-lime mortar strength test as method of comparing pozzolanic activity, (1955).
- [79] M. Collepardi, A. Marcialis, L. Massidda, U. Sanna, Low pressure steam curing of compacted limepozzolana mixtures, Cement and Concrete Research. 6 (1976) 497–506.
- [80] P. Hewlett, M. Liska, Lea's chemistry of cement and concrete, Butterworth-Heinemann, 2019.
- [81] C. Shi, R.L. Day, Chemical activation of blended cements made with lime and natural pozzolans, Cement and Concrete Research. 23 (1993) 1389–1396.
- [82] C. Shi, R.L. Day, Comparison of different methods for enhancing reactivity of pozzolans, Cement and Concrete Research. 31 (2001) 813–818.
- [83] G.M. Barger, A.J. Hulshizer, S. Popovics, B.M. Call, T.M. Jaber, J. Prusinski, R.L. Carrasquillo, J.S. Jensen,D. Ravina, J.E. Cook, Use of Raw or Processed Natural Pozzolans in Concrete, 2001.
- [84] K.M.A. Hossain, Blended cement using volcanic ash and pumice, Cement and Concrete Research. 33 (2003) 1601–1605.
- [85] M.J. Shannag, A. Yeginobali, Properties of pastes, mortars and concretes containing natural pozzolan, Cement and Concrete Research. 25 (1995) 647–657.

- [86] J.J. Brooks, M.M. Johari, Effect of metakaolin on creep and shrinkage of concrete, Cement and Concrete Composites. 23 (2001) 495–502.
- [87] R. Siddique, J. Klaus, Influence of metakaolin on the properties of mortar and concrete: A review, Applied Clay Science. 43 (2009) 392–400.
- [88] J.-I. Escalante-Garcia, J.H. Sharp, The chemical composition and microstructure of hydration products in blended cements, Cement and Concrete Composites. 26 (2004) 967–976.
- [89] J.I. Escalante-Garcia, J.H. Sharp, The microstructure and mechanical properties of blended cements hydrated at various temperatures, Cement and Concrete Research. 31 (2001) 695–702.
- [90] P. Bredy, M. Chabannet, J. Pera, Microstructure and porosity of metakaolin blended cements, MRS Online Proceedings Library Archive. 136 (1988).
- [91] J.M. Khatib, S. Wild, Pore size distribution of metakaolin paste, Cement and Concrete Research. 26 (1996) 1545–1553.
- [92] Z. Li, Z. Ding, Property improvement of Portland cement by incorporating with metakaolin and slag, Cement and Concrete Research. 33 (2003) 579–584.
- [93] C.-S. Poon, L. Lam, S.C. Kou, Y.-L. Wong, R. Wong, Rate of pozzolanic reaction of metakaolin in highperformance cement pastes, Cement and Concrete Research. 31 (2001) 1301–1306.
- [94] X. Jin, Z. Li, Effects of mineral admixture on properties of young concrete, Journal of Materials in Civil Engineering. 15 (2003) 435–442.
- [95] E. Badogiannis, G. Kakali, G. Dimopoulou, E. Chaniotakis, S. Tsivilis, Metakaolin as a main cement constituent. Exploitation of poor Greek kaolins, Cement and Concrete Composites. 27 (2005) 197–203.
- [96] N. Kaid, M. Cyr, S. Julien, H. Khelafi, Durability of concrete containing a natural pozzolan as defined by a performance-based approach, Construction and Building Materials. 23 (2009) 3457–3467.
- [97] T. Ramlochan, M. Thomas, K.A. Gruber, The effect of metakaolin on alkali–silica reaction in concrete, Cement and Concrete Research. 30 (2000) 339–344.
- [98] L. Courard, A. Darimont, M. Schouterden, F. Ferauche, X. Willem, R. Degeimbre, Durability of mortars modified with metakaolin, Cement and Concrete Research. 33 (2003) 1473–1479.
- [99] J.M. Khatib, R.M. Clay, Absorption characteristics of metakaolin concrete, Cement and Concrete Research. 34 (2004) 19–29.
- [100] M.H. Zhang, V.M. Malhotra, Characteristics of a thermally activated alumino-silicate pozzolanic material and its use in concrete, Cement and Concrete Research. 25 (1995) 1713–1725.

- [101] R.D. Hooton, K.A. Gruber, A. Boddy, The chloride penetration resistance of concrete containing highreactivity metakaolin, in: Proceedings of the PCI/FHWA International Symposium on High Performance Concrete, New Orleans, LA, 1997: pp. 172–183.
- [102] J. Bai, S. Wild, B.B. Sabir, Chloride ingress and strength loss in concrete with different PC-PFA-MK binder compositions exposed to synthetic seawater, Cement and Concrete Research. 33 (2003) 353–362.
- [103] C.-S. Poon, S.C. Kou, L. Lam, Compressive strength, chloride diffusivity and pore structure of high performance metakaolin and silica fume concrete, Construction and Building Materials. 20 (2006) 858–865.
- [104] G. Batis, P. Pantazopoulou, S. Tsivilis, E. Badogiannis, The effect of metakaolin on the corrosion behavior of cement mortars, Cement and Concrete Composites. 27 (2005) 125–130.
- [105] K.M.A. Hossain, M. Lachemi, Corrosion resistance and chloride diffusivity of volcanic ash blended cement mortar, Cement and Concrete Research. 34 (2004) 695–702.
- [106] K. Takewaka, Characteristics of Shirasu concrete: based on contents in the manual established by Kagoshima prefectural office, Concrete Journal. 45 (2007) 16–23.
- [107] D. Katpady, K. Takewaka, T.Yamaguchi. Experimental study on reactivity of pyroclastic flow deposit called Shirasu as binder in mortar and geopolymer. First International Conference on Concrete Sustainability, Tokyo 2013....
- [108] K. Takewaka, T.Yamaguchi, T.Nakazaki. Long term durability of reinforced concrete structures using "Shirasu concrete" in actual marine environment. The 3rd ACF international conference- ACF/VCA 2008...
- [109] D.Katpady, Experimental study on reactivity of pyroclastic flow deposit called shirasu and its practical use as cementitious material, doctoral thesis, Graduate School of Science and Engineering Kagoshima University, 2013.
- [110] B. Lothenbach, F. Winnefeld, C. Alder, E. Wieland, P. Lunk, Effect of temperature on the pore solution, microstructure and hydration products of Portland cement pastes, Cement and Concrete Research. 37 (2007) 483–491.
- [111] T. Bakharev, J.G. Sanjayan, Y.-B. Cheng, Effect of elevated temperature curing on properties of alkaliactivated slag concrete, Cement and Concrete Research. 29 (1999) 1619–1625.
- [112] A.A. Ramezanianpour, M.H. Khazali, P. Vosoughi, Effect of steam curing cycles on strength and durability of SCC: A case study in precast concrete, Construction and Building Materials. 49 (2013) 807–813.
- [113] Z. Zhang, B. Zhang, P. Yan, Hydration and microstructures of concrete containing raw or densified silica fume at different curing temperatures, Construction and Building Materials. 121 (2016) 483–490.

- [114] J.I. Escalante-Garcia, J.H. Sharp, The microstructure and mechanical properties of blended cements hydrated at various temperatures, Cement and Concrete Research. 31 (2001) 695–702.
- [115] T. Boubekeur, K. Ezziane, E.-H. Kadri, Estimation of mortars compressive strength at different curing temperature by the maturity method, Construction and Building Materials. 71 (2014) 299–307.
- [116] M. Ba, C. Qian, X. Guo, X. Han, Effects of steam curing on strength and porous structure of concrete with low water/binder ratio, Construction and Building Materials. 25 (2011) 123–128.
- [117] K. Ezziane, A. Bougara, A. Kadri, H. Khelafi, E. Kadri, Compressive strength of mortar containing natural pozzolan under various curing temperature, Cement and Concrete Composites. 29 (2007) 587–593.
- [118] S. Mattbäck, A. Boman, P. Österholm, Hydrogeochemical impact of coarse-grained post-glacial acid sulfate soil materials, Geoderma. 308 (2017) 291–301.
- [119] J. Monteny, E. Vincke, A. Beeldens, N. De Belie, L. Taerwe, D. Van Gemert, W. Verstraete, Chemical, microbiological, and in situ test methods for biogenic sulfuric acid corrosion of concrete, Cement and Concrete Research. 30 (2000) 623–634.
- [120] C.D. Parker, Species of sulphur bacteria associated with the corrosion of concrete, Nature. 159 (1947) 439.
- [121] C.D. Parker, The corrosion of concrete: 1. the isolation of a species of bacterium associated with the corrosion of concrete exposed to atmospheres containing hydrogen sulphide., Australian Journal of Experimental Biology and Medical Science. 23 (1945) 81–90.
- [122] V. Zivica, A. Bajza, Acidic attack of cement based materials—a review.: Part 1. Principle of acidic attack, Construction and Building Materials. 15 (2001) 331–340.
- [123] A. Ali, Š. František, Acidic corrosion of hydrated cement based materials: part 1-mechanism of the phenomenon, Ceramics-Silikáty. 44 (2000) 114–120.
- [124] E.K. Attiogbe, S.H. Rizkalla, Response of concrete to sulfuric acid attack, ACI Materials Journal. 85 (1988)481–488.
- [125] M.T. Bassuoni, M.L. Nehdi, Resistance of self-consolidating concrete to sulfuric acid attack with consecutive pH reduction, Cement and Concrete Research. 37 (2007) 1070–1084.
- [126] X. Shi, L. Fay, M.M. Peterson, M. Berry, M. Mooney, A FESEM/EDX investigation into how continuous deicer exposure affects the chemistry of Portland cement concrete, Construction and Building Materials. 25 (2011) 957–966.

- [127] L. Sutter, K. Peterson, S. Touton, T. Van Dam, D. Johnston, Petrographic evidence of calcium oxychloride formation in mortars exposed to magnesium chloride solution, Cement and Concrete Research. 36 (2006) 1533–1541.
- [128] M. Collepardi, L. Coppola, C. Pistolesi, Durability of concrete structures exposed to CaCl 2 based deicing salts, Special Publication. 145 (1994) 107–120.
- [129] Y. Farnam, T. Washington, J. Weiss, The influence of calcium chloride salt solution on the transport properties of cementitious materials, Advances in Civil Engineering. 2015 (2015).
- [130] D. Bonen, Composition and appearance of magnesium silicate hydrate and its relation to deterioration of cement-based materials, Journal of the American Ceramic Society. 75 (1992) 2904–2906.
- [131] L. Sutter, K. Peterson, G. Julio-Betancourt, D. Hooton, T.V. Dam, K. Smith, The deleterious chemical effects of concentrated deicing solutions on Portland cement concrete, South Dakota Department of Transportation, Office of Research, 2008.
- [132] V. Zivica, M.T. Palou, M. Krizma, L. Bagel, Acidic attack of cement based materials under the common action of high, ambient temperature and pressure, Construction and Building Materials. 36 (2012) 623–629.
- [133] G.L. Kalousek, A.F. Prebus, Crystal chemistry of hydrous calcium silicates: III, Morphology and other properties of tobermorite and related phases, Journal of the American Ceramic Society. 41 (1958) 124–132.
- [134] Y. Senhadji, G. Escadeillas, M. Mouli, H. Khelafi, Influence of natural pozzolan, silica fume and limestone fine on strength, acid resistance and microstructure of mortar, Powder Technology. 254 (2014) 314–323.
- [135] T.A. Durning, M.C. Hicks, Using microsilica to increase concrete's resistance to aggressive chemicals, Concrete International. 13 (1991) 42–48.
- [136] P. Jariyathitipong, K. Hosotani, T. Fujii, T. Ayano, Strength and durability of concrete with blast furnace slag, Proceedings of SCMT3. (2013) T4–6.
- [137] M.T. Bassuoni, M.L. Nehdi, Resistance of self-consolidating concrete to sulfuric acid attack with consecutive pH reduction, Cement and Concrete Research. 37 (2007) 1070–1084.
- [138] S. Aydın, H. Yazıcı, H. Yiğiter, B. Baradan, Sulfuric acid resistance of high-volume fly ash concrete, Building and Environment. 42 (2007) 717–721.
- [139] K. Torii, M. Kawamura, Effects of fly ash and silica fume on the resistance of mortar to sulfuric acid and sulfate attack, Cement and Concrete Research. 24 (1994) 361–370.

- [140] H. Siad, H.A. Mesbah, H. Khelafi, S. Kamali-Bernard, M. Mouli, Effect of mineral admixture on resistance to sulphuric and hydrochloric acid attacks in self-compacting concrete, Canadian Journal of Civil Engineering. 37 (2010) 441–449.
- [141] S. Türkel, B. Felekoğlu, S. Dulluc, Influence of various acids on the physico-mechanical properties of pozzolanic cement mortars, Sadhana. 32 (2007) 683–691.
- [142] D.M. Roy, P. Arjunan, M.R. Silsbee, Effect of silica fume, metakaolin, and low-calcium fly ash on chemical resistance of concrete, Cement and Concrete Research. 31 (2001) 1809–1813.
- [143] Z. Makhloufi, T. Bouziani, M. Hadjoudja, M. Bederina, Durability of limestone mortars based on quaternary binders subjected to sulfuric acid using drying–immersion cycles, Construction and Building Materials. 71 (2014) 579–588.
- [144] N. Kaid, M. Cyr, S. Julien, H. Khelafi, Durability of concrete containing a natural pozzolan as defined by a performance-based approach, Construction and Building Materials. 23 (2009) 3457–3467.
- [145] S. Cheng, Z. Shui, T. Sun, R. Yu, G. Zhang, Durability and microstructure of coral sand concrete incorporating supplementary cementitious materials, Construction and Building Materials. 171 (2018) 44– 53.
- [146] M. Ghrici, S. Kenai, E. Meziane, Mechanical and durability properties of cement mortar with Algerian natural pozzolana, Journal of Materials Science. 41 (2006) 6965–6972.
- [147] J. Schulze, Influence of water-cement ratio and cement content on the properties of polymer-modified mortars, Cement and Concrete Research. 29 (1999) 909–915.
- [148] G.A. Rao, Role of water-binder ratio on the strength development in mortars incorporated with silica fume, Cement and Concrete Research. 31 (2001) 443–447.

Chapter 3. EXPERIMENTAL PROGRAM

3.1. Introduction

This chapter presents the materials, mixture proportion, and experiments used in this study in detail. The physicochemical properties of limestone powder and Shirasu natural pozzolan, such as particle size distribution, the ignition loss, and density are mentioned in section 3.2. Section 3.3 explains the mixture proportion of various specimens.

Several experiments were conducted in the present study. Compression test was carried out to examine the compressive strength and the steady-state migration test to determine the chloride diffusion coefficient. The thermal analysis (TG-DTA) and chemical test were also carried out in order to determine the Portlandite (Ca (OH)₂) content of concrete mixtures under different curing temperatures. To investigate the deterioration mechanism of concretes caused by the aggressive environment, the immersion test was carried out. After 22 weeks of immersion, X-ray diffraction (XRD), compression test, erosion measurement, neutralization measurement and chloride penetration depth measurement were conducted. The concentration profile of ions was also established to determine the penetration level of SO_4^{2-} , Mg^{2+} , Cl⁻ ions in the concrete. Based on the experimental results, the mechanism of strength development and resistance of concrete to the attack of a mixed solution containing 3% H₂SO₄ and 5%MgCl₂ was investigated. The details of all these tests are explained in section 3.4.

3.2. Materials

The densities and chemical compositions of the cement, limestone powder and Shirasu natural pozzolan are presented in Table 3.1. Crushed quartz porphyry sand and stone were used as conventional fine and coarse aggregates, respectively. The water absorption values of the fine aggregate, coarse aggregate and Shirasu were 1.04%, 0.6% and 9.27%, respectively. The pozzolanic reactivity of the Shirasu was determined using the assessed pozzolanic-activity index (API) method [1] and found to be 14.4% compared to the value for commercial low-

calcium fly ash in the range of 25-85%. A water-reducing admixture that conformed to the JIS A 6204 standard (Chemical admixture for concrete) [2] was employed to achieve the desired workability for all concrete mixtures.

	Ordinary	Fine aggregate		
Chemical compositions /physical properties	Portland cement (C)	Limestone powder (L)	Natural pozzolan (NP)	Crushed quartz porphyry sand (S)
SiO ₂ (%)	20.29	0.19	73.61	75.63
Al ₂ O ₃ (%)	4.91	-	14.69	13.18
Fe ₂ O ₃ (%)	2.96	0.06	2.63	2.08
CaO (%)	65.05	55.75	2.02	1.07
MgO (%)	1.20	-	0.28	0.92
SO ₃ (%)	1.93	0.20	0.07	-
K ₂ O (%)	0.39	-	3.88	5.04
Na ₂ O (%)	0.26	-	2.27	1.54
Cl (%)	0.006	-	0.06	-
LOI (%)	2.44	43.91	2.39	0.60
Density (g/cm ³)	3.16	2.71	2.11	2.61
Fineness modulus	-	0.08	1.54	2.71

Table 3.1. Physical properties and chemical compositions of materials .

-: Not measured

3.3. Mixture proportions

The mixture proportions and the fresh properties of the concretes are summarized in Table 3.2. A reference concrete specimen (Ref) was made without replacing the fine aggregate. Herein, L5 and NP55 designate concretes in which 5% and 55% of the fine aggregate are replaced with limestone powder and Shirasu, respectively, while the L5NP55 is defined as the concrete with both 5% limestone powder and 55% Shirasu as fine aggregate replacements. In this study, limestone powder was used as a partial fine aggregate replacement due to its advantages based on filler, nucleation or chemical effects, and to mitigate disadvantages for the case of cement

replacement, such as the dilution effect [4]. The fine aggregate replacement proportion of 5% with limestone powder was based on previous studies [5–7]. The compressive strength and durability of concrete have been found to be enhanced when the mass percentage of limestone powder in the manufactured sand is less than 10-15% [5,6]. The replacement of 55% of the fine aggregate with the Shirasu natural pozzolan was based on the results of a previous study that used an Algerian natural pozzolan as a partial fine aggregate replacement in concrete [8]. Kaid et al. [8] reported that concrete using natural pozzolan as 20% of binder in addition to 13% of fine aggregate exhibited improved compressive strength after 28 days of cure and resistance to an ammonium acid solution (480 g/L).

The particle size distributions of the fine aggregates in each mixture are presented in Fig. 3.1. These data indicate that the aggregate mixtures containing Shirasu (NP55 and L5NP55) had greater proportions of fine particles with sizes of less than 0.3 mm compared to the formulation recommended by the JIS A 5005 standard (Crushed stone and manufactured sand for concrete) [9]. All fresh concrete mixtures in this study were designed to have a slump of 10.0 ± 2.0 cm and an air content of $2.0\pm0.5\%$.

		Unit	conte	ent (l	kg/m ³))		Slump (cm)		Air content (%)	
					Fin	e	Crushed	Design	Measured	Design	Measured
Mixture	W/C	С	W	;	aggreg	gate	quartz	value	value	value	value
		U	••	L	NP	S	porphyry				
							stone				
Ref	0.40	425	170	0	0	758	1005	10.0 ± 2.0	9.5	2.0±0.5	1.8
L5	0.40	425	170	39	0	720	1005	10.0±2.0	9.5	2.0±0.5	1.8
NP55	0.40	425	170	0	337	341	1005	10.0±2.0	9.5	2.0±0.5	2.0
L5NP55	0.40	425	170	39	337	303	1005	10.0±2.0	9.5	2.0±0.5	1.5

Table 3.2. Mixture proportions and fresh properties of the concrete specimens



Fig. 3.1. Particle size distributions of the fine aggregate mixtures .

3.4. Test procedures and measurements

3.4.1. Specimen preparation

Two types of specimens were produced: cylindrical specimens with 100 mm diameters and 200 mm heights for compressive strength measurement and cubic specimens with dimensions of $100 \times 100 \times 100$ mm for investigating the deterioration of concrete after immersing in the mixed solution via X-ray diffraction analysis, erosion depth, neutralization depth, chloride penetration depth measurements and concentration profile of ions establishment. After casting, specimens of each concrete mixture were cured at 20 °C while sealed with aluminum adhesive tape until tested to prevent water loss and carbonation.

3.4.2. Ca(OH)₂ content

At ages of 3, 28 and 91 days, the Ca(OH)₂ content in the cement pastes of the concrete specimens was ascertained by determining the cement content in each sample using a chemical test, combined with data from thermogravimetric and differential thermal analysis (TG-DTA). The samples for this test were obtained from the center part of concrete specimens following compressive strength tests. Each sample was soaked in acetone for 24 h to stop any further chemical reactions, after which the samples were dried in a vacuum desiccator for 24 h and then

crushed into fine powders (maximum particle size of 150 μ m) using a milling machine. The resulting homogenous powders were separated into two parts for the determination of cement content using a chemical test and the analysis of Ca(OH)₂ content based on TG-DTA.

The sodium gluconate dissolution method [10] was adopted for the determination of cement content. This test was performed by adding approximately 0.5 g concrete powder to 200 mL of a 15% sodium gluconate solution. The suspension was mixed using a magnetic stirrer at 300 rpm and gently boiled for 30 min. After stirring, the solution was passed through filter paper and the solid thus captured was dried at 105 °C for 12 h to eliminate free water, followed by cooling in a desiccator to ambient temperature for 30 min. Finally, the mass of the residue after drying was determined.

The Ca(OH)₂ content was found using TG–DTA, employing a DTG-60H instrument (Shimadzu Corporation). In each analysis, a sample of concrete powder was heated from room temperature to 100 °C at 20 °C/min, maintained at 100 °C for 30 min to remove free water, and then heated to 1000 °C at 20 °C/min. Based on the cement content results and the TG-DTA data, the Ca(OH)₂ content in the cement paste of each sample was determined using the equation

$$CH = CH_1 \times 100 / C,$$
 (3.1)

where CH is the $Ca(OH)_2$ content as a percentage of the cement paste (%), CH_1 is the $Ca(OH)_2$ content in the sample as determined by TG-DTA (%) and C is the cement paste content in the sample as calculated from the chemical test data (%). The last term is calculated as

$$C = (m_s - m_r) \times 100 / m_s, \tag{3.2}$$

where m_s is the mass of the concrete powder sample (g) and m_r is the mass of the residue after drying (g).

3.4.3. Compressive strength

At ages of 3, 28 and 91 days, the compressive strength of each specimen was measured according to the JIS A 1108 standard (Method of test for compressive strength of concrete) [11].

Three specimens were used for the compressive strength test at each age. The compression test was conducted at a constant load rate of 1.5 kN/s.

3.4.4. Chloride diffusion coefficient of concrete by steady-state migration test

The steady-state migration test [12,13] was adopted to find the diffusion coefficients of chloride ions in the concrete samples. Cylindrical concrete specimens with diameters of 100 mm and heights of 200 mm were fabricated and 30 mm thick slices were obtained from the center part of the specimens at the age of 91 days. The slice specimens were stored at 20° C and $60\pm5\%$ relative humidity for 4 h prior to coating the circumference of each specimen with epoxy resin and installing each sample in the migration test apparatus shown in Fig. 3.2. To establish a steady state, the concentration gradient must be constant during the test. Therefore, the solutions in the anode and cathode sections were renewed when the chloride ion concentration in the anode section was higher than 0.05 mol/L or that in the cathode was lower than 0.45 mol/L, based on the JSCE-G571 standard (Test method for effective diffusion coefficient of chloride ion in concrete by migration) [13]. Consequently, the solutions in the anode sections were renewed after approximately 185 h of migration. Effective chloride diffusion coefficients for the L5, NP55 and LNP55 were determined based on three concrete discs while the value for the Ref mixture was obtained using two specimens because of limited experimental equipment. A voltage of 12 V was applied between the two sides of each concrete sample, although the actual measured potential was found to be approximately 10 V, and so this value was used when calculating the diffusion coefficients. All tests were carried out at 20±2 °C. The chloride concentration was measured by ion chromatography using a Dionex ICS-1000 instrument. The flux of chloride ions in the steady state was calculated using the equation

$$J_{Cl} = \frac{V}{A} \frac{\Delta C_{Cl}}{\Delta t},\tag{3.3}$$

where J_{Cl} is the flux of chloride ions in the steady state (mol/m²s), *V* is the volume of solution in the anode compartment (m³), A is the cross-sectional area of the concrete sample (m²) and $\Delta C_{Cl}/\Delta_t$ is the increase in concentration of chloride ions in the anode (mol/m³/s). The effective chloride diffusion coefficient was calculated using the equation

$$D_e = \frac{J_{Cl}RTL}{|Z|EFC_{Cl}},\tag{3.4}$$

where D_e is the effective diffusion coefficient (m²/s), R is the gas constant (8.3144 J/mol K), T is the absolute temperature (K), Z is the charge on a chloride ion (-1), F is the Faraday constant (96,480 J/V mol), C_{Cl} is the average chloride ion concentration in the cathode (mol/m³), E is the electrical potential difference between specimen surfaces (V) and L is the length of the specimen (m).



Fig. 3.2. The apparatus used for steady-state migration experiments.

3.4.5. Immersion test

3.4.5.1. Procedures

After being cured for 56 days at 20 °C while sealed with aluminum adhesive tape, the concrete specimens were immersed in a mixed solution containing 3% H₂SO₄ and 5% MgCl₂. Prior to immersion, all but one designated surface of each cubic specimen was coated with an acid resistant epoxy resin. In the case of the cylindrical specimens, the two surfaces at both ends of the specimen were coated with this resin. The mixed solution was refreshed two weeks after immersion and every four weeks thereafter so as to maintain a pH in the range of 0.3 to 0.7 for
a period of 22 weeks, at a temperature of 20 ± 3 °C. The ratio of the volume of the mixed solution to the surface area of the specimen was fixed at approximately 8 mL/cm² to maintain constant contact conditions between the specimen and the solution. Several specimens were also immersed in deionized water for comparison purposes.

3.4.5.2. X-ray diffraction (XRD) analysis

Dried powdered samples were extracted from the outermost layer of the surfaces of concrete specimens immersed in the mixed solution or in deionized water for 22 weeks. These powders were assessed by XRD to identify the primary crystalline phases, using an AXS D_2 Phaser X-ray diffractometer (Bruker Corporation). Patterns were obtained over the 2 θ range from 5° to 65° using a Cu K α X-ray source and the data were analyzed using the EVA software program.

3.4.5.3. Erosion depth

A typical erosion depth measurement is shown in Fig. 3.3a. After the specimen was removed from the mixed solution and blotted with paper towels, the length of the specimen was determined using an electronic caliper with 0.01 mm accuracy at six points on the sample. The erosion depth was then calculated using the equation

Erosion depth (mm) =
$$H_0 - H_t$$
, (3.5)

where H_o is the original length of the specimen before immersion (mm) and H_t is the length of the specimen after immersion (mm). Each specimen was tested in triplicate and the mean values are reported. It should be noted that in the case that the sample expanded, the erosion depth would be negative.



Fig. 3.3. Erosion depth (a), chloride penetration (b) and depth neutralization depth (c) measurements

3.4.5.4. Compressive strength loss

The ASTM C267-01 standard [14] (Test method for chemical resistance of mortars, grouts, and monolithic surfacing and polymer concretes) was used to determine the compressive strength of concrete specimens after immersion in the mixed solution. After 22 weeks of immersion, each specimen was extracted from the solution and blotted with paper towels. The elapsed time between the removal of the specimen from the solution and the compressive strength test was 60 min, at 20°C. The residual compressive strength of the specimen was subsequently determined, using the original cross-sectional area of the specimen before immersion. For each concrete mixture, three specimens were used for the compressive strength test. The strength loss was determined using the equation

Compressive strength loss (%) = $(C_w - C_m) \times 100 / C_w$, (3.6)

where C_m is an average of three compression test results for specimens immersed in the mixed solution for a period of time, t (N/mm²) and C_w is an average of three compression test results for specimens immersed in deionized water for the same period of time (N/mm²).

3.4.5.5. Neutralization depth

An example of the measurement of the neutralization depth of a sample after immersion in the mixed solution is presented in Fig. 3.3 (c). After being taken out of the solution and blotted with paper towels, each cubic specimen was cut into two at its center and the neutralization depth

was detected by spraying the freshly split surface with a 1% phenolphthalein solution. The depth was ascertained at nine positions on the sample using an electronic caliper with 0.01 mm accuracy. The neutralization depth was then calculated using the equation

Neutralization depth (mm) =
$$N_w + E$$
, (3.7)

where N_w is the depth of the uncolored zone after spraying with the phenolphthalein solution (mm) and E is the erosion depth determined using Equation (5) (mm).

3.4.5.6. Chloride penetration depth measurement

The chloride penetration depth of each specimen was measured by spraying silver nitrate indicator. After being taken out of solution and blotted with paper towels, the cubic specimens were split into two parts at the middle height of the specimens. The chloride penetration depth was detected by spraying 0.1M silver nitrate solution on the freshly split surface of concretes. The chloride penetration depth was measured by using an electronic caliper with 0.01 mm accuracy. The chloride penetration depth determined by Equation (17) was calculated as the mean measured values of nine positions on the split surface of concretes.

Chloride penetration depth (mm) =
$$CP_d + E$$
 (3.8)

where CP_d: the depth of grey zone after spraying silver nitrate solution (mm).

E: erosion depth in Equation (5) (mm).

The measurement of chloride penetration depth of concrete immersed in the mixed solution was shown in Fig.3.3(b). In Fig.3.3(b), CP_o is the original depth of specimen before immersion in the mixed solution.

3.4.5.7. Concentration profiles of magnesium, chloride and sulfate ions in concretes

In preparation for analyses, a powdered sample was collected from the neutralized region of each specimen by abrading the specimen with a handheld mini grinder from the surface to the depth at which the neutralization frontier disappeared (as determined based on the brown coloration between the neutralized and non-neutralized regions; see Fig. 3.4). In addition, slices

were cut from the non-neutralized zone using an oil saw at 5 mm intervals, as shown in Fig. 3.4. After oil on the surfaces of the slices was blotted with paper towels, the slices were soaked in acetone for 30 min to completely remove residual oil. Then, the mortar samples from slices were collected and dried in a desiccator for 24 h before being ground into powders less than 150 μm in size. Approximately 1 g of each powder sample was poured into 50 mL of a 2 M HNO₃ solution, after which the solution was covered with a watch glass and stirred magnetically for approximately 30 min at 300 rpm and 100 °C. Subsequently, the suspension was cooled to ambient temperature and filtered, following which the concentration of magnesium ions in the filtrate was determined using a Shimazu AA-6800 atomic absorption spectrophotometer. The concentrations of chloride and sulfate ions were determined using a Dionex ICS-1000 ion chromatography system based on the JIS A1154 standard [15] and a previous report [16].

The concentrations of Mg^{2+} , Cl^- and SO_4^{2-} ions in concrete specimens immersed in deionized water were also ascertained using mortar samples collected from the center part of each specimen.



Fig. 3.4. Sampling of a deteriorated concrete specimen to measure Cl^{-} , SO_4^{2-} and Mg^{2+} ion

concentrations

References in chapter 3

- T. Yamamoto, T. Kanazu, M. Nambu, T. Tanosaki, Pozzolanic reactivity of fly ash–API method and Kvalue, Fuel. 85 (2006) 2345–2351.
- [2] JIS A 6204 Chemical admixture for concrete, Japanese Standards Association, Tokyo (2011)..
- [3] Van Toan Pham, Phalkong Meng, Phuong Trinh Bui, Yuko Ogawa, Kenji Kawai, Effects of Shirasu natural pozzolan and limestone powder on the strength and aggressive chemical resistance of concrete, Construction and Building Materials, accepted in 2019.
- [4] D. Wang, C. Shi, N. Farzadnia, Z. Shi, H. Jia, Z. Ou, A review on use of limestone powder in cement-based materials: Mechanism, hydration and microstructures, Construction and Building Materials. 181 (2018) 659–672.
- [5] B. Li, J. Wang, M. Zhou, Effect of limestone fines content in manufactured sand on durability of low- and high-strength concretes, Construction and Building Materials. 23 (2009) 2846–2850.
- [6] H. Yang, D. Liang, Z. Deng, Y. Qin, Effect of limestone powder in manufactured sand on the hydration products and microstructure of recycled aggregate concrete, Construction and Building Materials. 188 (2018) 1045–1049.
- T. Celik, K. Marar, Effects of crushed stone dust on some properties of concrete, Cement and Concrete Research. 26 (1996) 1121–1130.
- [8] N. Kaid, M. Cyr, S. Julien, H. Khelafi, Durability of concrete containing a natural pozzolan as defined by a performance-based approach, Construction and Building Materials. 23 (2009) 3457–3467.
- [9] JIS A 5005 Crush stone and manufactured sand for concrete, Japanese Standards Association, Tokyo (2009).
- [10] Y. Kasai, K. Matsui, Study on determination of cement content test method with sodium gluconate. Review of the 40th general meeting/ Technical session, pp. 214-217, 1986 (in Japanese).
- [11] JIS A 1108 Method of test for compressive strength of concrete, Japanese Standards Association, Tokyo (2006).
- [12] O. Truc, J.P. Ollivier, M. Carcassès, A new way for determining the chloride diffusion coefficient in concrete from steady state migration test, Cement and Concrete Research. 30 (2000) 217–226.
- [13] JSCE-G571 Test method for effective diffusion coefficient of chloride ion in concrete by migration, Japan Society of Civil Engineers, Tokyo (2003).
- [14] ASTM C267-01, Standard test methods for chemical resistance of mortars, grouts, and monolithic surfacing and polymer concretes, (2012).

- [15] JIS A1154: Concentration of chloride ions contained in hardened concrete test method, Japanese Standard Association, Tokyo (2011).
- [16] T.C. Rilem, Analysis of total chloride content in concrete; RILEM TC-TMC: Testing and modelling chloride penetration in concrete, Materials and Structure. 38 (2012) 583–855.

Chapter 4. EFFECTS OF LIMESTONE POWDER AND SHIRASU NATURAL POZZOLAN ON THE STRENGTH DEVELOPMENT AND CHEMICAL RESISTANCE OF CONCRETE

4.1. Introduction

This chapter discusses the effects of limestone powder and Shirasu natural pozzolan on strength development and resistance of concrete to the attack of the mixed solution containing 3% sulfuric acid and 5% magnesium chloride which simulate the aggressive chemical environment. As mentioned in chapter 1, there are a few studies concerning the effects of limestone powder and natural pozzolan used as partial fine aggregate replacement on the strength development and resistance of concrete to the aggressive chemical environment containing H⁺, SO₄²⁻, Mg²⁺ and Cl⁻ ions. While, H⁺, SO₄²⁻, Mg²⁺ and Cl⁻ ions are considered to be the most destructive ions for concrete and can deteriorate its service life and increase the maintenance cost of civil infrastructures. Based on the experimental results, the effect of limestone powder and Shirasu natural pozzolan on the strength development and chemical resistance of concrete are analyzed and explained in this chapter. The deterioration mechanism of concrete caused by the attack of aggressive chemical environment containing H⁺, SO₄²⁻, Mg²⁺ and Cl⁻ ions is also discussed.

4.2. Experiments

4.2.1. Mixture proportion

Four mixture proportion was produced by using limestone powder and Shirasu natural pozzolan as fine aggregate replacement. The proportions of fine aggregate replaced with the limestone powder and Shirasu were 5% and 55% by volume, respectively. A constant water/cement ratio of 0.40 and a cement content of 425 kg/m³ were used for all concrete mixtures. The mixture proportions and the fresh properties of the concretes are summarized in Table 4.1. A reference concrete specimen (Ref) was made without replacing the fine aggregate.

	W/C	Unit	Unit content (kg/m ³)					Slump (cm)		Air content (%)	
Mixture		С	W	Fine aggregate		Crushed quartz	Design value	Measured value	Design value	Measured value	
				L	NP	S	porphyry stone				
Ref	0.40	425	170	0	0	758	1005	10.0±2.0	9.5	2.0 ± 0.5	1.8
L5	0.40	425	170	39	0	720	1005	10.0±2.0	9.5	2.0±0.5	1.8
NP55	0.40	425	170	0	337	341	1005	10.0±2.0	9.5	2.0±0.5	2.0
L5NP55	0.40	425	170	39	337	303	1005	10.0±2.0	9.5	2.0±0.5	1.5

Table 4.1. Mixture proportions and fresh properties of the concrete specimens

4.2.2. Specimen preparation and curing condition

Two types of specimens were produced: cylindrical specimens with 100 mm diameters and 200 mm heights for compressive strength measurement and cubic specimens with dimensions of $100 \times 100 \times 100$ mm for the other measurements. After casting, specimens of each concrete mixture were cured at 20 °C while sealed with aluminum adhesive tape until tested to prevent water loss and carbonation

4.2.3. Test procedures and measurements

In this study, the thermal analysis (TG-DTA) and chemical test were also carried out in order to determine the Portlandite (Ca(OH)₂) content of concrete mixtures. To investigate the deterioration mechanism of concretes caused by the aggressive environment, the immersion test was carried out. After 22 weeks of immersion, X-ray diffraction (XRD), compression test, erosion measurement was conducted. The concentration profile of ions was also established to determine the penetration level of SO_4^{2-} , Mg^{2+} , Cl⁻ ions in the concrete. The details of these tests were mentioned in chapter 3.

4.3. Results and discussion

4.3.1. Ca(OH)₂ content

Figure 4.1 summarizes the Ca(OH)₂ contents in concrete specimens following aging for 3, 28 and 91 days. It can be seen that, after 3 days, the Ca(OH)₂ contents in the pastes extracted from

concretes both with Shirasu (L5NP55 and NP55) and without (Ref and L5) were almost equal. However, the Ca(OH)₂ contents in the L5NP55 and NP55 specimens were significantly lower than those in the Ref and L5 following 28 and 91 days. These reductions in the Ca(OH)₂ content after prolonged aging can be attributed to the pozzolanic reactivity of the Shirasu, which consumes Ca(OH)₂ from the cement paste to produce additional cementitious materials such as calcium silicate hydrates (C–S–H) or calcium aluminum silicate hydrates (C–A–S–H) [1]. Thus, pozzolanic reactions evidently proceeded in specimens in which 55% of the fine aggregate was replaced by Shirasu during a 28-day curing under sealed conditions at 20 °C. In addition, the Ca(OH)₂ content in the L5 specimen was very similar to that of the Ref specimen regardless of curing time, indicating that the effect of limestone powder on the cement hydration process was insignificant.



Fig. 4.1. Ca(OH)₂ content in concrete samples at different ages

4.3.2. Compressive strength

The compressive strengths of the concrete specimens at different curing ages are presented in Fig. 4.2. At the age of 3 days, the compressive strengths of the specimens containing Shirasu, including the NP55 and L5NP55, were almost similar to those of the Ref and L5. The effect of adding Shirasu on strength development during this time frame was therefore insignificant under sealed curing conditions at 20 °C. However, at the later ages (i.e. at 28 and 91 days), the compressive strengths of the NP55 and L5NP55 were considerably higher than those of the

concretes Ref and L5. Considering the Ca(OH)₂ data, it is apparent that the pozzolanic reaction between the Shirasu and Ca(OH)₂ produced additional hydration products, resulting in a refinement effect and increased strength after 28 days of curing [1–4]. The compressive strengths of the NP55 and L5NP55 were almost the same regardless of the aging period. In addition, the compressive strength of the L5 containing limestone powder as a partial fine aggregate replacement was close to that of the Ref specimen after aging for 3, 28 and 91 days. These results indicate that the effect of the limestone powder on the strength development of the concrete was negligible. In contrast, using the Shirasu natural pozzolan as 55% of fine aggregate replacement improved the compressive strength.



Fig. 4.2. Compressive strengths of concrete specimens after aging for 3, 28 and 91 days

4.3.3. Immersion test

4.3.3.1. XRD analysis

The main crystalline phases of concretes immersed in the mixed solution or in deionized water were identified by XRD. The resulting patterns are provided in Fig. 4.3. These data indicate that all specimens immersed in deionized water produced similar peaks associated with primary phases of portlandite (Ca(OH)₂) or calcite (CaCO₃). As an example, the prominent peaks at 2θ values of 18.1° , 34.2° and 47.3° demonstrate the presence of Ca(OH)₂. However, there was no trace of Ca(OH)₂ in the patterns of concretes immersed in the mixed solution. These results revealed that the Ca(OH)₂ was completely decomposed due to chemical reactions with the H₂SO₄ and MgCl₂. The intense peaks related to CaCO₃ also disappeared in the patterns of those concretes immersed in the mixed solution, indicating that Ca(OH)₂ and CaCO₃ are vulnerable to chemical reaction with the mixed solution. These results also imply that the presence of Ca(OH)₂ as well as CaCO₃, a major constituent of limestone powder, resulted in more severe deterioration of the L5 immersed in the mixed solution. The patterns also display prominent peaks related to bassanite (CaSO₄·0.5H₂O), gypsum (CaSO₄·2H₂O), epsomite (MgSO₄·7H₂O) and hydrophilite (CaCl₂) in the case of specimens immersed in the mixed solution.

The main phases detected by the XRD analysis are summarized in Table 4.2. These data confirm that the chemical reactions between the mixed solution and cementitious materials such as Ca(OH)₂ and C–S–H resulted in a large number of reaction products, including CaSO₄·0.5H₂O, CaSO₄·2H₂O, MgSO₄·7H₂O and CaCl₂ which are non-cementitious materials. Such reactions led to strength loss and disintegration of the concrete. Reaction products such as gypsum (CaSO₄·2H₂O) associated with volume expansion can also precipitate and fill in the pores of the concrete, resulting in internal stress and cracking [5–10].



Fig. 4.3. XRD patterns of concretes immersed in the mixed solution or in deionized water for

Phase	Formula	Peak at 20	Remarks		
Quartz	SiO	20.9°, 26.6°,			
	5102	50.2°			
Gungum	C.SO. 2H.O	11.7°, 20.7°,	These peaks were more prominent in samples		
Gypsum	CaSO4.2H ₂ O	29.1°	immersed in the mixed solution.		
		14.8°, 25.7°,	These nears were only detected in complex		
Bassanite	CaSO ₄ .0.5H ₂ O	29.8°, 31.9°,	These peaks were only detected in samples		
		49.3°	immersed in the mixed solution.		
	Ca(OH) ₂	18.1. 34.2°.	These peaks were more prominent in all		
Portlandite		47 3°	mixtures cured in water and disappeared in		
		т.5	all mixtures immersed in the mixed solution.		
		20 10 30 10	These peaks were more prominent in L5		
Calcite	CaCO ₃	29.4, 39.4,	mixture cured in water and disappeared in L5		
		43.1°,	mixtures immersed in the mixed solution.		
Encomito	Maso, 7 Hao	22.50	This peak was only detected in samples		
Epsonne	wig504.7 m20	33.3	immersed in the mixed solution.		
Hudronhilita	CaCla	19.8°, 29.3°,	This peak was only detected in samples		
Hydrophilite		38.6°	immersed in the mixed solution.		

Table 4.2. The main phases in concretes immersed in deionized water and the mixed solution

detected by XRD method

4.3.3.2. Erosion depth

The erosion depths of specimens after immersion in the mixed solution are shown in Fig. 4.4. Here, a positive value of erosion depth indicates actual erosion of the specimen while a negative value demonstrates expansion. These results show that expansion occurred primarily during the initial 10 weeks of immersion for all specimens. After 12 weeks, the Ref and L5 concretes exhibited significant erosion, while the NP55 and L5NP55 showed slight erosion. After 22 weeks of immersion, the erosion depths of concretes containing Shirasu, including the NP55 and L5NP55, were lower than those of the Ref and L5. These results indicate that the addition of Shirasu slowed the erosion process. The XRD results confirmed the generation of chemical reaction products in all specimens, implying that the observed initial expansion can be attributed to the formation of products such as gypsum or bassanite [5-10]. However, prolonged immersion resulted in disintegration of the cementitious materials and reaction products due to attack by aggressive chemicals in the mixed solution, leading to dramatic erosion.

The lower erosion depths of the specimens containing the Shirasu (L5NP55 and NP55) may be attributed to the limited Ca(OH)₂ contents in the mixtures due to the consumption of Ca(OH)₂ through the pozzolanic reactions with the Shirasu. It is known that among the hydration products, Ca(OH)₂ is the most likely to react with acids [11–13]. Furthermore, the addition of Shirasu may beneficially affect the chemical resistance of the concrete. Prior work has shown that the C–S–H produced by pozzolanic reactions has a denser structure and lower CaO/SiO₂ ratio than conventional C–S–H produced by ordinary Portland cement [5,14]. Chatveera et al. [15] also found that both the chemical composition and CaO/SiO₂ ratio of hydrated cement paste are important factors that determine the resistance to acids. It has also been reported that C–S–H having a high CaO/SiO₂ ratio preferentially undergoes decalcification and is susceptible to further acid attack, while C–S–H with a low CaO/SiO₂ ratio degrades at a slower rate [5,16].



Fig. 4.4. Erosion depths of concrete specimens during immersion in the mixed solution

4.3.3.3. Compressive strength loss

The compressive strengths of concrete specimens immersed in the mixed solution or deionized water for a period of 22 weeks are summarized in Table 4.3. It can be seen that the residual compressive strengths of the specimens containing the Shirasu (NP55 and L5NP55) were significantly higher than those of concretes without the Shirasu (Ref and L5). In addition, in the case of immersion in deionized water, the compressive strengths of the former concretes were higher than those of the latter concretes because of the pozzolanic reactivity of Shirasu (as discussed in Section 4.3.2).

Table 4.3. Compressive strengths of concrete specimens after 22 weeks of immersion in

 deionized water or the mixed solution along with standard deviations (S.D.)

Mixture	Compressive strength (N/mm ²)						
	Deionized water	S.D.	Mixed solution	S.D.			
Ref	64.0	1.2	38.2	1.3			
L5	64.5	2.5	37.7	3.6			
NP55	70.5	1.5	46.1	2.8			
L5NP55	68.0	0.4	43.6	0.6			

Figure 4.5 shows the compressive strength losses of the concretes after 22 weeks of immersion in the mixed solution, as calculated using Equation (3.6). These data indicate that the highest compressive strength loss in the L5. In contrast, the best resistance to the aggressive chemical solution was found in the case of the NP55. The Ca(OH)₂ content of concretes had a major effect on the deterioration rate, which is associated with the residual strength [11–13]. Thus, the lower deterioration rates of the NP55 and L5NP55 can be attributed to the reduced Ca(OH)₂ contents in concrete containing Shirasu due to the pozzolanic reactivity of the Shirasu, as seen in Fig. 4.1. In addition, the pozzolanic reactions of the Shirasu resulted in the formation of additional C–S–H and C–A–S–H, leading to refinement of the pore structure [1–4]. This microstructure refinement effect prevented the diffusion of ions into the concrete, leading to reduced deterioration of the concretes. The lower compressive strength losses of the NP55 and L5NP55 are also due to changes in the chemical composition of the hydrated cement paste resulting from the formation of secondary hydration products after the pozzolanic reactions. The pozzolanic reaction of natural pozzolan generates additional calcium-aluminate hydrates that are more stable in highly acidic environments than calcium silicate hydrates [17,18]. In contrast, the greater compressive strength loss of the L5 compared to the Ref specimen can be attributed to the higher CaCO₃ content in the former, since CaCO₃ will react with the chemicals in the mixed solution, as was determined from the XRD analysis. The resulting reaction products, such as gypsum, are typically associated with the volume expansion of concretes, which in turn can form micro cracks [5–10]. This phenomenon leads to deeper diffusion of ions and thus an increased the deterioration rate of concrete.



Fig. 4.5. Compressive strength losses of concrete specimens after 22 weeks of immersion

4.3.3.4. Concentration profiles of magnesium, chloride and sulfate ions in concretes.

The penetration profiles of Cl⁻, SO_4^{2-} and Mg^{2+} ions for the various concretes are presented in Figs. 4.6, 4.7 and 4.8, respectively. According to Fig. 4.6, the concentration of Cl⁻ ions decreased with increasing depth from the exposed surface. The chloride penetration depth was approximately 18.5 mm for the specimens containing the Shirasu (NP55 and L5NP55), which is less than the depths of approximately 24.5 mm for the L5 and approximately 29.5 mm for the Ref. These results indicate that using Shirasu as a partial fine aggregate replacement improved the resistance to Cl⁻ ion penetration. Figures 4.7 and 4.8 demonstrate that the concentrations of

 SO_4^{2-} and Mg^{2+} ions in concretes immersed in the mixed solution were almost the same as those in concretes immersed in deionized water, except at the shallowest depths. These results suggest that the penetration of SO_4^{2-} and Mg^{2+} ions was essentially stopped at the frontier between the neutralized and non-neutralized zones, as shown in Fig. 4 in chapter 3. Thus, the behaviors of these ions differed from that of the Cl⁻ ions, which passed the neutralization frontier. The deeper penetration of the latter ions is attributed to the more rapid diffusion of Cl⁻ ions. These results are in agreement with some previous studies [19–22]. Obserholster [19] reported that the diffusion of Cl⁻ ions is 10-100 times faster than that of SO_4^{2-} ions in concrete, while Stratful [21] determined that the diffusion rate of Cl⁻ ions was 14 times higher. Bakker [22] reported that the diffusion rate of Cl⁻ ions in hardened cement pastes is significantly faster than those of SO_4^{2-} and other cations.

In this study, the diffusion coefficients of ions were determined by fitting the ion content plots via a least-squares method using the function

$$C(x,t) = C_o + (C_s - C_o) \cdot \left(1 - erf\left(\frac{x}{2 \cdot \sqrt{D_c \cdot t}}\right) \right)$$
(5)

where C(x,t) is the ion concentration at depth x and time t (%), C_o is the ion concentration of the concrete prior to immersion (%), C_s is the ion concentration at the surface (%) and D_C is the apparent diffusion coefficient of the ions (m²/s). The apparent diffusion coefficients (D_C) of the various ions are presented in Table 4.

The results in Table 4.4 demonstrate that the apparent diffusion coefficients of chloride and sulfate ions in the concretes containing Shirasu, including the NP55 and L5NP55, were lower than those for the concretes without Shirasu (Ref and L5), while the effect of Shirasu on the apparent diffusion coefficient of magnesium ions was not significant. In addition, the effect of limestone powder on the apparent diffusion coefficient of each ion was insignificant. In general, the apparent diffusion coefficient of chloride ions was the highest, and that of magnesium ions

was almost equal to that of sulfate ions. In this study, the diffusion coefficient of chloride ions was 6-13 times higher than that of sulfate or magnesium ions.

Table 4.4. Apparent diffusion coefficients, D_C , of ions after 22 weeks of immersion in the mixed solution

Mintune	Apparent	t diffusion coefficient, I	$D_{\rm C} ({\rm m}^2/{\rm s})$
WIXture	Cl-	SO4 ²⁻	Mg^{2+}
Ref	3.69 ×10 ⁻¹²	2.80 ×10 ⁻¹³	3.02 ×10 ⁻¹³
L5	3.79 ×10 ⁻¹²	2.83 ×10 ⁻¹³	2.83 ×10 ⁻¹³
NP55	1.61 ×10 ⁻¹²	2.51 ×10 ⁻¹³	2.89 ×10 ⁻¹³
L5NP55	2.17 ×10 ⁻¹²	2.59 ×10 ⁻¹³	2.96 ×10 ⁻¹³



Fig. 4.6. The Cl⁻ ion penetration profiles in various concretes after 22 weeks of immersion



Fig. 4.7. The SO_4^{2-} ion penetration profiles in various concretes after 22 weeks of immersion



Fig. 4.8. The Mg^{2+} ion penetration profiles in various concretes after 22 weeks of immersion

4.4. Conclusions

This study analyses effects of limestone powder and Shirasu natural pozzolan on the strength development and chemical resistance of concretes in detail. The following conclusions can be drawn based on the mechanical and chemical analyses in this work:

- The usage of Shirasu as a fine aggregate replacement improved the compressive strength of concrete due to its pozzolanic reactivity.
- The effect of limestone powder on strength development was insignificant.

• The addition of Shirasu as a partial fine aggregate replacement improved the resistance of the concrete to an aggressive chemical solution containing 3% H₂SO₄ and 5% MgCl₂.

• The concrete using limestone powder as a partial fine aggregate replacement was suffered more severe deterioration of concretes by the aggressive chemical solution. However, the incorporation of the Shirasu compensated for this effect.

References in chapter 4

[1] A.M. Neville, Properties of Concrete. (5th edition), Addison Wesley, Longman, England (1996).

[2] K. Ezziane, A. Bougara, A. Kadri, H. Khelafi, E. Kadri, Compressive strength of mortar containing natural pozzolan under various curing temperature, Cement and Concrete Composites. 29 (2007) 587–593.

[3] B. Uzal, L. Turanlı, Blended cements containing high volume of natural zeolites: Properties, hydration and paste microstructure, Cement and Concrete Composites. 34 (2012) 101–109.

[4] L. Bertolini, M. Carsana, M. Frassoni, M. Gelli, Pozzolanic additions for durability of concrete structures, Proceedings of the Institution of Civil Engineers-Construction Materials. 164 (2011) 283–291.

[5] M.T. Bassuoni, M.L. Nehdi, Resistance of self-consolidating concrete to sulfuric acid attack with consecutive pH reduction, Cement and Concrete Research. 37 (2007) 1070–1084.

[6] K. Kawai, A. Sako, T. Ikuta, T. Ishida., Deterioration of cement hydrates containing mineral admixture due to sulfuric acid attack. 11DBMC International conference on durability of building materials and components, Turkey (2008)

[7] I. Kurashige, T. Uomoto, Deterioration of concrete due to sulfuric acid, Transactions of the Japan Concrete
 Institute. 22 (2001) 167–174.

[8] D. Benavente, M.G. del Cura, J. Garcia-Guinea, S. Sánchez-Moral, S. Ordóñez, Role of pore structure in salt crystallisation in unsaturated porous stone, Journal of Crystal Growth. 260 (2004) 532–544.

[9] D. Benavente, J. Martínez-Martínez, N. Cueto, M.A. García-del-Cura, Salt weathering in dual-porosity building dolostones, Engineering Geology. 94 (2007) 215–226.

[10] E. Ruiz-Agudo, F. Mees, P. Jacobs, C. Rodriguez-Navarro, The role of saline solution properties on porous limestone salt weathering by magnesium and sodium sulfates, Environmental Geology. 52 (2007) 269–281.

 [11] V. Zivica, A. Bajza, Acidic attack of cement based materials—a review.: Part 1. Principle of acidic attack, Construction and Building Materials. 15 (2001) 331–340. [12] A. Ali, Š. František, Acidic corrosion of hydrated cement based materials: part 1-mechanism of the phenomenon, Ceramics-Silikáty. 44 (2000) 114–120.

[13] H. Siad, M. Lachemi, M. Sahmaran, K.M.A. Hossain, Effect of glass powder on sulfuric acid resistance of cementitious materials, Construction and Building Materials. 113 (2016) 163–173.

[14] R.J. Detwiler, J.I. Bhatty, S. Battacharja, Supplementary cementing materials for use in blended cements, 1996.

[15] B. Chatveera, P. Lertwattanaruk, Durability of conventional concretes containing black rice husk ash, Journal of Environmental Management. 92 (2011) 59–66.

[16] C. Shi, J.A. Stegemann, Acid corrosion resistance of different cementing materials, Cement and Concrete Research. 30 (2000) 803–808.

[17] Y. Senhadji, G. Escadeillas, M. Mouli, H. Khelafi, Influence of natural pozzolan, silica fume and limestone fine on strength, acid resistance and microstructure of mortar, Powder Technology. 254 (2014) 314–323.

[18] V. Zivica, A. Bajza, Acidic attack of cement-based materials—a review Part 2. Factors of rate of acidic attack and protective measures, Construction and Building Materials. 16 (2002) 215–222.

[19] R.E. Oberholster, Pore structure, permeability and diffusivity of hardened cement paste and concrete in relation to durability: status and prospects, in: 8th International Congress on Chemistry of Cement, 1986: pp. 323–335.

[20] H.A. Dehwah, M. Maslehuddin, S.A. Austin, Long-term effect of sulfate ions and associated cation type on chloride-induced reinforcement corrosion in Portland cement concretes, Cement and Concrete Composites. 24 (2002) 17–25.

[21] R.F. Stratful, Effect on reinforced concrete in sodium chloride and sodium sulfate environments, Materials Protection. 3 (1964) 74–80.

[22] R.F. Bakker, Permeability of blended cement concretes, ACI-SP. 79 (1983) 589-606.

Chapter 5. EFFECTS OF INITIAL CURING TEMPERATURE ON THE STRENGTH DEVELOPMENT AND CHEMICAL RESISTANCE OF CONCRETE CONTAINING LIMESTONE POWDER AND SHIRASU NATURAL POZZOLAN

5.1. Introduction

This chapter discusses the effects of initial curing temperature on the compressive strength development and chemical resistance of concrete containing limestone powder and Shirasu natural pozzolan. As mentioned in chapter 1, many previous studies reported that curing temperature significantly influence the hydration process of cement as well as the pozzolanic reaction [1–6]. However, there are a few studies concerning effects of initial curing temperature on the compressive strength development and chemical resistance of concrete containing limestone powder and Shirasu natural pozzolan to the aggressive chemical environment containing H⁺, SO4²⁻, Mg²⁺ and Cl⁻ ions. While, H⁺, SO4²⁻, Mg²⁺ and Cl⁻ ions are considered to be the most destructive ions to concrete and can deteriorate its service life and increase the maintenance cost of civil infrastructures. Based on the results of the experimental results, the effect of initial curing temperature on the compressive strength development and Shirasu natural pozzolan was analyzed and explained.

5.2. Experiments

5.2.1. Mixture proportion

Three mixture proportion was produced by using limestone powder and Shirasu natural pozzolan as fine aggregate replacement. The proportions of fine aggregate replaced with the limestone powder and Shirasu were 5% and 55% by volume, respectively. A constant water/cement ratio of 0.40 and a cement content of 425 kg/m³ were used for all concrete mixtures. The mixture proportions and the fresh properties of the concretes are summarized in Table 5.1. A reference concrete specimen (Ref) was made without replacing the fine aggregate.

	W/C	Unit content (kg/m ³)						Slump (cm)		Air content (%)	
Mixture			W/		Fin	e	Crushed	Design	Measured	Design	Measured
		C		;	aggreg	gate	quartz	value	value	value	value
		C	vv	L	NP	S	porphyry				
							stone				
Ref	0.40	425	170	0	0	758	1005	10.0±2.0	9.5	2.0±0.5	1.8
L5	0.40	425	170	39	0	720	1005	10.0±2.0	9.5	2.0±0.5	1.8
NP55	0.40	425	170	39	337	303	1005	10.0±2.0	9.5	2.0±0.5	2.0

Table 5.1. Mixture proportions and fresh properties of the concrete specimens.

5.2.2. Specimen preparation and curing condition

Two types of specimens were produced: one was cylindrical specimens of 100 mm diameter and 200 mm height for compression tests and other was cubic specimen with the size of 100×100×100 mm for other tests. After casting, specimens of each concrete mixture were separated into two groups: one was applied for normal curing temperature condition in which specimens were cured in a sealed condition with aluminum tape at 20°C until the designated test ages (i.e., 3, 28, and 91 days for the compressive strength measurements); whereas the other was placed under sealed condition with aluminum tape at high curing temperature history of 60°C for 8 hours followed by cooling process in the curing chamber. The schematic of heat treatment regime which was referred to the curing temperature history of the previous studies [3,4,7] is shown in Fig. 5.1. In the previous studies, this heat treatment regime was applied for investigating the effects of steam curing on the properties of self-compacting concrete containing limestone powder which is used in the precast concrete plant [3] and recycled aggregate fly ash concrete [4,7]. In the study, this method was considered also for examining the effect of initial curing temperature on the properties of concrete containing Shirasu natural pozzolan and limestone powder in this study. After the initial curing stage of 24 hours, the specimens were cured under sealed condition with aluminum tape at 20°C until the designated test ages



Fig. 5.1. Schematic representation of the heat treatment procedure

5.2.3. Test procedures and measurements

In this study, the compression test was conducted for determination of strength development of concrete. The thermal analysis (TG-DTA) and chemical test were also carried out in order to analysis the Portlandite (Ca(OH)₂) content of concrete mixtures. To investigate the chloride resistance of concrete, the steady-state migration test [8] was adopted to find the chloride diffusion coefficients. The deterioration mechanism of concretes cured at different initial curing temperatures caused by the aggressive environment was also studied. The immersion test was carried out. After 22 weeks of immersion, X-ray diffraction (XRD), compression test, the measurement of erosion depth, neutralization depth, chloride penetration depth was conducted. The detail of these tests was mentioned in chapter 3.

5.3. Results and discussion

5.3.1. Ca(OH)₂ content

Figure 5.2 shows the Ca(OH)₂ content of concrete samples cured at different temperatures. It can be seen that under normal curing temperature, the Ca(OH)₂ contents in concretes containing Shirasu (NP55) substantially reduced at the ages of 28 and 91 days in comparison with that in concretes without Shirasu (Ref and L5). It is assumed that the consumption of Ca(OH)₂ owing to the pozzolanic reaction between Shirasu and Ca(OH)₂ led to lower Ca(OH)₂ content in the concrete containing Shirasu than that in concretes without Shirasu.

The Ca(OH)₂ content in NP55 was significantly lower than that in Ref cured at high temperature condition regardless of the ages. Especially, at the age of 3-day curing, the Ca(OH)₂ contents in concretes containing Shirasu (NP55) was almost equal to that of concrete without Shirasu under normal curing temperature, however, at the same time under high curing temperature, the Ca(OH)₂ contents of NP55 was significant lower than that of Ref and L5 concretes. It can be inferred that elevated temperature can promote the pozzolanic reactivity of Shirasu, which accelerates the consumption of Ca(OH)₂ at the early ages.

In addition, the use of limestone as fine aggregate replacement insignificantly affected the hydration process of cement pastes regardless of the curing temperature. As a result, the Ca(OH)₂ content of L5 was almost identical to that of Ref regardless of curing temperature conditions at different curing ages.





5.3.2. Compressive strength

The compressive strength of concrete specimens at different curing ages is illustrated in Fig. 5.3. It can be observed that under normal curing condition, at the age of 3 days, the compressive strength of concrete containing Shirasu (NP55) is almost the same as those of concrete without Shirasu (Ref and L5). However, at the later age, the compressive strength of NP55 is significantly higher than that of concrete without Shirasu after 28 and 91 days under normal

curing temperature conditions. Considering the results of the Ca(OH)₂ content as shown in Fig. 5.2, it can be said that the pozzolanic reaction between Shirasu and Ca(OH)₂ produced additional cementitious hydration products, namely calcium silicate hydrates (C-S-H) and calcium aluminate hydrates (C-A-H) and refined the pore structure of cement matrix [9], leading to the strength development of concretes. Furthermore, Fig. 3 also shows that under high curing temperature condition, the compressive strength of concrete containing Shirasu (NP55) moderately exceeds those of concretes without Shirasu (Ref and L5) at the age of 3 days. It indicates that initial high curing temperature can promote the pozzolanic reactivity of Shirasu, leading to the formation of additional cementitious hydration products and the increase of the compressive strength of concrete containing Shirasu at the early age. Considering the results of the Ca(OH)₂ content as shown in Fig. 2, at the age of 3 days, under high curing temperature condition, the Ca(OH)₂ content of concrete containing Shirasu was significantly lower than those of concretes without Shirasu due to the pozzolanic reactivity of Shirasu. In the case of L5 concrete, the effect of limestone powder on the compressive strength of

concrete was insignificant. The strength of L5 was almost the same as that of Ref regardless of curing temperatures.

In addition, the compressive strength at early age (i.e., at the age of 3 days) of concretes cured at high temperature was higher than that of concretes cured at normal temperature. On the other hand, the later-age compressive strength of all concrete mixtures cured at high temperature condition were less than that of concrete cured under normal curing condition after 28 and 91 days of curing. This result agreed with the previous studies [10–12]. It is reported that concrete exposed to elevated temperature shows accelerated cement hydration, leading to the increase in compressive strength of concrete at the early age. However, more precipitated hydration products that quickly deposited may form a barrier for further ion diffusion and hydration

process, resulting in inhomogeneity of microstructure and significant strength reduction of concretes at the later age [10,11].



Fig. 5.3. Compressive strengths of concrete specimens after aging for 3, 28 and 91 days.

5.3.3. Chloride diffusion coefficient

Figures 5.4 and 5.5 show the total increase of chloride concentration in the anode solution over time and the chloride diffusion coefficient of mixture proportions, respectively. It indicates that the lowest rate of chloride diffusion was recorded for concretes containing Shirasu (NP55) regardless of curing temperature condition, followed by the rate of chloride diffusion of L5. Ref concrete without replacing the fine aggregate showed the highest rate of chloride diffusion. Obviously, the addition of Shirasu improved the resistance of concrete to chloride permeability because pozzolanic reaction led to the formation of additional cementitious material which contributed to refinement pore structure and the reduction of interconnection of pores [13]. This result also agreed with the report of [14] on the resistance of concrete using Shirasu as fine aggregate to the chloride-induced deterioration during 3 years of immersion in the tidal sea zone. In addition, using limestone powder as fine aggregate replacement also reduced the chloride permeability of concretes. It can be explained that the filling effect of limestone powder in the voids between aggregates and cement grains could result in the densification of concrete structure and contribute to impede the diffusion of ions from external environments into concretes.

Furthermore, the effective chloride diffusion coefficients of concretes cured at normal temperature were lower than those of concretes cured at high temperature, as seen in Fig. 8. It is assumed that higher porosity of concretes cured at high temperature [11,15] resulted in higher chloride permeability than that of concretes cured at normal temperature. Detwiler et al. [16] also reported that elevated curing temperatures result in coarser pore structure and a corresponding decrease in the resistance to chloride ion diffusion.



Fig. 5.4. Cumulative increase of chloride concentration in anode sections.



Fig. 5.5. Chloride diffusion coefficients of concretes

5.3.4. Immersion test

5.3.4.1. XRD analysis

Figure 5.6 shows the patterns of the major peaks detected by the XRD test in the specimens immersed in the mixed solution and pure water. It indicates that all samples immersed in deionized water showed prominent peaks for the main phases of portlandite (Ca(OH)₂). However, it can be seen that no peak of Ca(OH)₂ was observed in the patterns of concrete immersed in the mixed solution. In addition, high peaks of calcite ($CaCO_3$) appeared in L5 sample immersed in water. Nevertheless, CaCO₃ peaks disappeared from all patterns of concretes immersed in the mixed solution. It indicates that Ca(OH)₂ and CaCO₃ are most susceptible to react with the mixed solution containing sulfuric acid and magnesium chloride. The presences of high contents of Ca(OH)₂ and Ca(OH)₂ were responsible for more severe deterioration of L5 concrete than other concretes immersed in the mixed solution. In addition, prominent peaks of bassanite (CaSO₄.0.5H₂O), gypsum (CaSO₄.2H₂O), hydrophilite (CaCl₂) and epsomite (MgSO₄.7 H₂O) observed in the XRD patterns of all concretes immersed in the mixed solution confirmed the formation of these reaction products caused by attack of the mixed solution. These products are non-cementitious materials which lead to strength loss of concrete structure. In addition, reaction products such as gypsum (CaSO₄.2H₂O) associated with volume expansion could precipitate and fill in the pores of concrete structure, leading to the internal stress and crack in concrete structure, eventually [17,18].



B: Bassanite (CaSO₄.0.5H₂O), C: Calcite (CaCO₃), E: Epsomita (MgSO₄.7 H₂O), G: Gypsum (CaSO₄.2H₂O), H: Hydrophilite (CaCl₂), P: Portlandite (Ca(OH)₂), Q: Quartz (SiO₂).

Fig. 5.6. XRD patterns of concretes after 22 weeks of immersion in water and mixed solution. Name of specimens labelled with N - normal curing temperature; H - high curing temperature.

5.3.4.2. Erosion depth

The erosion depths of specimens immersed in the mixed solution is shown in Fig. 5.7. In the Fig. 5.7, a negative value of erosion depth implies the expansion of specimen while a positive value of erosion depth indicates the erosion of specimen. The results show that the expansion process was almost observed during the initial 10 weeks of immersion in the mixed solution for all mixtures. After 12 weeks, the erosion significantly occurred in cases of Ref and L5 concretes. while, the NP55 showed a slight erosion. In addition, after 22 weeks, the erosion depths of concretes containing Shirasu (NP55) were lower than those of Ref and L5 regardless of curing temperature conditions. It indicates that the addition of Shirasu mitigated the erosion of the concretes immersed in the mixed solution. Considering the XRD results, the generation of chemical reaction products, including gypsum (CaSO4.2H₂O), bassanite (CaSO4.0.5H₂O), epsomite (MgSO4.7 H₂O), hydrophilite (CaCl₂) was confirmed. It implies that the occurrence of expansion was attributed to the formation of reaction products such as gypsum (CaSO4.2H₂O) which is usually associated with volume expansion of concretes [20–23].

However, long-term immersion, concrete was decomposed by aggressive chemical reaction between the mixed solution and cementitious materials, leading to serious erosion. The decomposition of concrete structure can be confirmed by the evolution of reaction products such as gypsum (CaSO₄.2H₂O), bassanite (CaSO₄.0.5H₂O), epsomite (MgSO₄.7 H₂O), hydrophilite (CaCl₂).

The lower erosion depth of concrete containing Shirasu in comparison to that of concrete without Shirasu may be attributed to lower Ca(OH)₂ content which is available to react with sulfuric acid and magnesium chloride. Furthermore, the addition of Shirasu natural pozzolan may influence on the chemical resistance of concrete. Detwiler et al. [24] pointed out that the C-S-H produced by pozzolanic reaction has a denser structure and calcium-to-silicate ratio lower than conventional C-S-H produced by OPC hydration. It was also pointed out that the lower C/S ratio and Ca(OH)₂ content contribute to reduction of acidic deterioration rate of concrete [20,22,23].

Moreover, the erosion depth of concretes cured at high temperature was lower than that of concretes cured at normal temperature because of lower Ca(OH)₂ content as seen in Fig. 2. In addition, the precipitated deposition of chemical reaction products in the voids of concrete structure acted as a barrier to slow down the attack of aggressive chemicals [17,20,25]. It is assumed that the concretes cured at high temperature have higher porosity [1,12,26] leading to the larger precipitated reaction products in the pore structure as a barrier to reduce the deterioration rate of concretes cured at high temperature in comparison with that of concretes cured at normal temperature.



Fig. 5.7. Erosion depth of concrete after 22 weeks of immersion in the mixed solution.

5.3.4.3. Compressive strength loss

After 22 weeks of immersion in the mixed solution, the residual compressive strength of the specimens was determined. The compressive strength of concrete specimens immersed in the mixed solution and deionized water for a period of 22 weeks are listed in Table 3. It can be seen that the residual compressive strength of concrete containing Shirasu (NP55) was higher than that of other mixtures without Shirasu (Ref and L5) at the same immersion period regardless of the curing temperature.

Figure 5.8 shows the compressive strength loss of concretes after 22 weeks of immersion in the mixed solution. It indicates that the highest compressive strength loss was recorded for the mixture containing limestone powder (L5). In contrast, the lowest compressive strength loss was found to be in NP55 compared to other concretes in both curing temperature conditions. The reason for the higher resistance of NP55 in comparison with the other concretes may be the lower $Ca(OH)_2$ content, which is the most vulnerable to aggressive chemical attack. On the other hand, L5 contained high amounts of $Ca(OH)_2$ and $CaCO_3$ which were vulnerable to chemical reaction as mentioned in the XRD analysis, resulting in the highest compressive

strength loss. In addition, the pozzolanic reaction of Shirasu resulted in the formation of additional calcium silicate hydrate (C-S-H) and calcium aluminum silicate hydrate (C-A-S-H) [9]. The additional hydration products lead to refinement effect of the pore structure, including reduction in interconnectivity of the pore structure [27]. The refinement effect prevented the diffusion of H^+ , SO_4^{2-} , Mg^{2+} and Cl^- ions into the concrete structure.

Moreover, after the 22-week immersion, the strength loss of all concretes cured at normal temperature was considerably higher than that of specimens cured at high temperature conditions. Regarding the mechanism of concrete deterioration caused by attack of a high concentration of sulfuric acid solution at pH degree less than 1, previous studies reported that sulfuric acid reacts with cement hydration products to produce expansive reaction products such as gypsum, leading to increase in volume of solid substances which causes expansion, internal stress and cracks in the concrete structure [17,18]. In addition, concrete with higher amount of larger capillary pores showed higher absorption capacity of expansion derived from the deposition of reaction products, leading to less severe micro crack-induced deterioration than that with lower amount of large capillary pores [28,29]. It was reported that after immersion in high concentration of sulfuric acid solution, more cracks can be occurred in concretes with lower W/C ratio associated with lower porosity than in concretes with higher W/C ratio.

For the aforementioned reasons, including lower Ca(OH)₂ content and higher expansion absorption capacity associated with less cracks in concrete structure, the concretes cured at high temperature showed a lower deterioration rate than the deterioration rate of concretes cured at normal temperature.

 Table 5.3. Compressive strength of concrete specimens after 22-week immersion in deionized

 water and the mixed solution and the standard deviation of compressive strength values (S.D.)

Curing	Mixture	Compressive strength (N/mm ²)				
condition		pure water	S.D.	mixed	S.D.	
				solution		

Normal	N-Ref	64.0	1.2	38.2	1.3
	N-L5	64.5	2.5	37.7	3.6
lemperature	N-NP55	70.5	1.5	46.1	2.8
II: al	H-Ref	61.5	0.4	39.4	1.1
temperature	H-L5	62.2	1.6	39.0	3.5
	H-NP55	57.1	1.1	39.9	0.6



Fig. 5.8. Compressive strength loss of concrete after 22 weeks of immersion in the mixed solution.

5.3.4.4. Neutralization depth

The neutralization depth of concretes after 22 weeks of immersion in the mixed solution is illustrated in Fig. 5.9. It can be seen that the highest neutralization depth was observed in L5 regardless of curing temperature conditions. The lowest neutralization depth was recorded for NP55. It indicates that the addition of Shirasu enhanced the resistance of concrete to the mixed solution attack owing to the pozzolanic reactivity of Shirasu associated with lower Ca(OH)₂ content and refinement effect on concrete structure. In contrast, using limestone powder as fine aggregate replacement adversely affected the resistance of concretes because of higher content of Ca(OH)₂ and CaCO₃ which is vulnerable to the chemical reaction as mentioned in the XRD analysis. Furthermore, the lower neutralization depth was recorded for concretes cured at high

temperature in comparison with that of concretes cured at normal temperature because of lower portlandite content and coarser porosity associated with higher absorption capacity of expansion caused by the precipitation of chemical reaction products such as bassanite (CaSO4.0.5H₂O), gypsum (CaSO4.2H₂O), hydrophilite (CaCl₂) and epsomite (MgSO4.7 H₂O).





5.3.4.5. Chloride penetration depth

The chloride penetration depth in concrete after 22 weeks of immersion in the mixed solution is presented in Fig. 5.10. It can be seen that concrete using Shirasu as a fine aggregate replacement (NP55) exhibited the lower neutralization depth in comparison with other concretes without Shirasu (Ref and L5). It confirms that Shirasu can enhance the resistance of concrete to the chloride diffusion because of its pozzolanic reactivity. The lower chloride penetration of concrete containing Shirasu could be attributed to pore structure refinement, resulting from the additional hydration products formed by the pozzolanic reaction [30]. The refinement effects might result in the decrease in the connectivity of pores and contribute to impeding the diffusion of ions from external environments into concretes. In addition, chloride penetration depth in concrete containing limestone powder (L5) was lower than that in Ref. It indicates that the addition of limestone powder as a fine aggregate replacement can improve the chloride impermeability of concretes. It is assumed that the filling effect of limestone powder in the voids between aggregates and cement grains could result in the densification of concretes structure.

Furthermore, Fig. 5.10 shows that the chloride penetration depth in concretes cured at a high temperature is higher than that in concretes cured at a normal temperature. It confirms that the elevated curing temperature can result in high chloride permeability of concretes due to higher porosity of concrete structure caused by initial high curing temperature [31,32].



Fig. 5.10. Chloride penetration depth of concretes after 22 weeks of immersion in the mixed

solution

5.4. Conclusions

This study analyses effects of initial curing temperature on the strength development and chemical resistance of concretes limestone powder and Shirasu natural pozzolan. The following conclusions can be drawn based on the mechanical and chemical analyses in this work:

• The high initial curing temperature can accelerate the pozzolanic reactivity of Shirasu, leading to an increase the compressive strength of concretes containing Shirasu as a partial fine aggregate replacement at the early age. However, initial high curing temperature tends to reduce the compressive strength of concrete.

• The deterioration of concretes cured at high initial temperature by the attack of a mixed solution containing 3% sulfuric acid and 5% magnesium chloride was less severe than that of concretes cured at normal temperature. On the other hand, high curing temperature can result in higher chloride penetration depth of concretes in comparison with that of concretes cured at normal temperature.

References in chapter 5

- B. Lothenbach, F. Winnefeld, C. Alder, E. Wieland, P. Lunk, Effect of temperature on the pore solution, microstructure and hydration products of Portland cement pastes, Cement and Concrete Research. 37 (2007) 483–491.
- [2] T. Bakharev, J.G. Sanjayan, Y.-B. Cheng, Effect of elevated temperature curing on properties of alkaliactivated slag concrete, Cement and Concrete Research. 29 (1999) 1619–1625.
- [3] A.A. Ramezanianpour, M.H. Khazali, P. Vosoughi, Effect of steam curing cycles on strength and durability of SCC: A case study in precast concrete, Construction and Building Materials. 49 (2013) 807–813.
- [4] A. Gonzalez-Corominas, M. Etxeberria, C.S. Poon, Influence of steam curing on the pore structures and mechanical properties of fly-ash high performance concrete prepared with recycled aggregates, Cement and Concrete Composites. 71 (2016) 77–84.
- [5] Z. Zhang, B. Zhang, P. Yan, Hydration and microstructures of concrete containing raw or densified silica fume at different curing temperatures, Construction and Building Materials. 121 (2016) 483–490.
- [6] K. Ezziane, A. Bougara, A. Kadri, H. Khelafi, E. Kadri, Compressive strength of mortar containing natural pozzolan under various curing temperature, Cement and Concrete Composites. 29 (2007) 587–593.
- [7] S.C. Kou, C.S. Poon, D. Chan, Properties of steam cured recycled aggregate fly ash concrete, in: International RILEM Conference on the Use of Recycled Materials in Buildings and Structures, Barcelona, 2004: pp. 590–599.
- [8] O. Truc, J.P. Ollivier, M. Carcassès, A new way for determining the chloride diffusion coefficient in concrete from steady state migration test, Cement and Concrete Research. 30 (2000) 217–226.
- [9] A.M. Neville, Properties of concrete, Longman London, 1995.
- [10] T. Bakharev, J.G. Sanjayan, Y.-B. Cheng, Effect of elevated temperature curing on properties of alkaliactivated slag concrete, Cement and Concrete Research. 29 (1999) 1619–1625.
- [11] K. Ezziane, A. Bougara, A. Kadri, H. Khelafi, E. Kadri, Compressive strength of mortar containing natural pozzolan under various curing temperature, Cement and Concrete Composites. 29 (2007) 587–593.
- [12] J. Paya, J. Monzo, M.V. Borrachero, E. Peris-Mora, F. Amahjour, Mechanical treatment of fly ashes: Part IV. Strength development of ground fly ash-cement mortars cured at different temperatures, Cement and Concrete Research. 30 (2000) 543–551.
- [13] B. Uzal, L. Turanlı, Blended cements containing high volume of natural zeolites: Properties, hydration and paste microstructure, Cement and Concrete Composites. 34 (2012) 101–109.
- [14] K. Takewaka, T. Yamaguchi, T. Nakazaki, Long-term durability of reinforced concrete structures using "shirasu concrete" in an actual marine environment, (2008).
- [15] B. Lothenbach, F. Winnefeld, C. Alder, E. Wieland, P. Lunk, Effect of temperature on the pore solution, microstructure and hydration products of Portland cement pastes, Cement and Concrete Research. 37 (2007) 483–491.
- [16] R.J. Detwiler, K.O. Kjellsen, O.E. Gjorv, Resistance to chloride intrusion of concrete cured at different temperatures, Materials Journal. 88 (1991) 19–24.
- [17] A. Allahverdi, F. Škvára, Acidic corrosion of hydrated cement based materials, Ceramics- Silikáty. 44 (2000) 152–160.
- [18] V. Zivica, A. Bajza, Acidic attack of cement based materials—a review.: Part 1. Principle of acidic attack, Construction and Building Materials. 15 (2001) 331–340.
- [19] Van Toan Pham, Phalkong Meng, Phuong Trinh Bui, Yuko Ogawa, Kenji Kawai, Effects of Shirasu natural pozzolan and limestone powder on the strength and aggressive chemical resistance of concrete, Construction and Building Materials, 239 (2020) 117679.
- [20] M.T. Bassuoni, M.L. Nehdi, Resistance of self-consolidating concrete to sulfuric acid attack with consecutive pH reduction, Cement and Concrete Research. 37 (2007) 1070–1084.
- [21] K. Kawai, S. Yamaji, T. Shinmi, Concrete deterioration caused by sulfuric acid attack, in: International Conference on Durability of Building Materials and Components, 2005: pp. 17–20.
- [22] Y. Senhadji, G. Escadeillas, M. Mouli, H. Khelafi, Influence of natural pozzolan, silica fume and limestone fine on strength, acid resistance and microstructure of mortar, Powder Technology. 254 (2014) 314–323.
- [23] H. Siad, M. Lachemi, M. Sahmaran, K.M.A. Hossain, Effect of glass powder on sulfuric acid resistance of cementitious materials, Construction and Building Materials. 113 (2016) 163–173.

- [24] R.J. Detwiler, J.I. Bhatty, S. Battacharja, Supplementary cementing materials for use in blended cements, 1996.
- [25] Z. Makhloufi, T. Bouziani, M. Hadjoudja, M. Bederina, Durability of limestone mortars based on quaternary binders subjected to sulfuric acid using drying–immersion cycles, Construction and Building Materials. 71 (2014) 579–588.
- [26] W.H. Price, Factors influencing concrete strength, in: Journal Proceedings, 1951: pp. 417–432.
- [27] B. Uzal, L. Turanlı, Blended cements containing high volume of natural zeolites: Properties, hydration and paste microstructure, Cement and Concrete Composites. 34 (2012) 101–109.
- [28] K. Kawai, A. Sako, T. Ikuta, T. Ishida., Deterioration of cement hydrates containing mineral admixture due to sulfuric acid attack. 11DBMC International conference on durability of building materials and components, Turkey (2008).
- [29] I. Kurashige, T. Uomoto, Deterioration of concrete due to sulfuric acid, Transactions of the Japan Concrete Institute. 22 (2001) 167–174.
- [30] A.M. Neville, Properties of concrete, Longman London, 1995.
- [31] B. Lothenbach, F. Winnefeld, C. Alder, E. Wieland, P. Lunk, Effect of temperature on the pore solution, microstructure and hydration products of Portland cement pastes, Cement and Concrete Research. 37 (2007) 483–491.
- [32] K. Ezziane, A. Bougara, A. Kadri, H. Khelafi, E. Kadri, Compressive strength of mortar containing natural pozzolan under various curing temperature, Cement and Concrete Composites. 29 (2007) 587–593.

Chapter 6. CONCLUSIONS AND RECOMMENDATIONS

6.1. Conclusions

This study analyses effects of limestone powder and Shirasu natural pozzolan on the strength development and chemical resistance of concrete to the attack of aggressive chemical environment containing H⁺, SO4²⁻, Mg²⁺, Cl⁻ ions which are considered to be the most destructive ions to concrete and can deteriorate its service life and increase the maintenance cost of civil infrastructure. In addition, the effect of initial curing temperature on the strength development and chemical resistance of concretes limestone powder and Shirasu natural pozzolan was also studied. In the study, limestone powder and Shirasu natural pozzolan was used as partial fine aggregate replacement. The proportions of fine aggregate replaced with the limestone powder and Shirasu were 5% and 55% by volume, respectively. After being cured for 56 days at 20 °C while sealed with aluminum adhesive tape, the concrete specimens were immersed in a mixed solution containing 3% H₂SO₄ and 5% MgCl₂. Based on the mechanical and chemical analyses, the following conclusions can be drawn in this work:

(1) Effects of limestone powder and Shirasu natural pozzolan

• The addition of Shirasu as a fine aggregate replacement improved the compressive strength of concrete due to its pozzolanic reactivity.

• The effect of limestone powder on strength development was insignificant.

• The addition of Shirasu as a partial fine aggregate replacement improved the resistance of the concrete to an aggressive chemical solution containing 3% H₂SO₄ and 5% MgCl₂.

• The concrete using limestone powder as a partial fine aggregate replacement was suffered more severe deterioration of concretes by the aggressive chemical solution. However, the incorporation of the Shirasu compensated for this effect.

(2) Effects of initial curing temperature

• The high initial curing temperature can accelerate the pozzolanic reactivity of Shirasu, leading to an increase the compressive strength of concretes containing Shirasu as a partial fine aggregate replacement at the early age. However, initial high curing temperature tends to reduce the compressive strength of concrete.

• The deterioration of concretes cured at high initial temperature by the attack of a mixed solution containing 3% sulfuric acid and 5% magnesium chloride was less severe than that of concretes cured at normal temperature. On the other hand, high curing temperature can result in higher chloride penetration depth of concretes in comparison with that of concretes cured at normal temperature.

(3) The deterioration mechanism of concretes caused by the attack of aggressive environment containing 3% sulfuric acid and 5% magnesium chloride

• The deterioration of concrete caused by aggressive chemical containing sulfuric acid and magnesium chloride solution was confirmed by the decomposition of hydration products, calcite of hardened concrete and formation of reaction products, such as gypsum (CaSO₄.2H₂O), bassanite (CaSO₄.0.5 H₂O), epsomite (MgSO₄.7 H₂O), and hydrophilite (CaCl₂), silica gel (SiO₂). Herein, reaction products such as gypsum (CaSO₄·2H₂O) associated with volume expansion can also precipitate and fill in the pores of the concrete, resulting in internal stress and cracking of the concrete structure.

6.2. Recommendations for future study

The present study investigated the effects of limestone powder and Shirasu natural pozzolan on the strength development and chemical resistance of concrete cured at the different initial temperatures. Some recommendations for future work can be made in detail as follows:

• In the study, only the content of limestone powder of 5% and Shirasu of 55% as partial fine aggregate replacements were designated based on those reported in some previous studies. It is

necessary to extend for various amounts of content of limestone powder and Shirasu should be tested to determine at what point the addition of limestone and Shirasu stop being effective.

• In the study, a limited test for investigating the effect of limestone powder and Shirasu were carried out. It is necessary to extend for other tests, such as elastic modulus, tensile strength, water permeability, carbonation, alkali-silica reaction, etc.

• Because of the limited time, an accelerated test with the use of a mixed solution containing 3% sulfuric acid and 5% magnesium chloride was used for simulating the attack of an aggressive chemical environment. Therefore, other test solutions with various concentrations of sulfuric acid and magnesium chloride should be done in order to simulate various conditions in the site. In addition, other tests using the original environment, such as water in acid sulfate soil areas, should be conducted in further study to simulate the real condition.

• Regarding investigation of temperature on the performance of concrete, a high curing temperature history of 60°C for 8 hours was adopted for specimens in the present study. Other temperatures curing regimes should be applied to obtain the optimum curing temperature for the performance of concretes contaning limestone powder and Shirasu.

• In addition, the other curing conditions that can affect the mechanical properties and durability of concrete should be adopted in the future works, such as steam curing or drying condition without sealed aluminum tape in the air at ambient temperature.

Appendix: Publications arising from this research

List of papers were published or in the process of publishing from this research:

A. Academic Journals

1) Van Toan Pham, Phalkong Meng, Phuong Trinh Bui, Yuko Ogawa, Kenji Kawai, Effects of shirasu natural pozzolan and limestone powder on the strength and aggressive chemical resistance of concrete, Construction and Building Materials, Vol. 239 117679 online (2020). DOI: 10.1016/j.conbuildmat.2019.117679.

2) Toan V. Pham, Meng Phalkong, Yuko Ogawa, Kenji Kawai, Deterioration of concretes containing Shirasu natural pozzolan and limestone powder exposed to aggressive chemical environment, Proceeding of the Japan Concrete Institute. Vol. 41, No.1, (2019), pp. 647-652.

3) V.T. Pham, M. Phalkong, Y. Ogawa, K. Kawai, Effects of Shirasu Volcanic Ash and Limestone Powder on Chemical Resistance of Concrete, Proceeding of the 8th International Conference of Asian Concrete Federation, Vol. 1 (2018), pp. 483-493.

B. Presentations at International Conferences

1) V.T. Pham, M. Phalkong, Y. Ogawa, K. Kawai, Effects of shirasu volcanic ash and limestone powder on chemical resistance of concrete, The 8th International Conference of Asian Concrete Federation – ACF 2018, Fuzhou, China (2018).

2) Toan V. Pham, Meng Phalkong, Yuko Ogawa, Kenji Kawai, Deterioration of concrete containing Shirasu natural pozzolan and limestone powder exposed to aggressive chemical environment, Japan Concrete Institute - JCI Annual Convention 2019, Sapporo, Japan (2019).